

Received 28 March 2024, accepted 16 April 2024, date of publication 22 April 2024, date of current version 29 April 2024. Digital Object Identifier 10.1109/ACCESS.2024.3391945

TOPICAL REVIEW

Advancements and Challenges in Scalable Modular Antenna Arrays for 5G Massive MIMO Networks

ARSELAN ASHRAF^{®1}, (Student Member, IEEE), TEDDY SURYA GUNAWAN^{®1,2}, (Senior Member, IEEE), MIRA KARTIWI^{®3}, (Member, IEEE), LEVY OLIVIA NUR^{®2}, (Member, IEEE), BAMBANG SETIA NUGROHO², (Member, IEEE), AND RINA PUDJI ASTUTI², (Member, IEEE)

¹Department of Electrical and Computer Engineering, International Islamic University Malaysia (IIUM), Kuala Lumpur 53100, Malaysia ²School of Electrical Engineering, Telkom University, Bandung 40257, Indonesia

³Information Systems Department, International Islamic University Malaysia (IIUM), Kuala Lumpur 53100, Malaysia

Corresponding author: Teddy Surya Gunawan (tsgunawan@iium.edu.my)

This work was supported by Telkom University under Grant KWR4.044/PNLT2/PPM-LIT/2021.

ABSTRACT This paper presents a comprehensive review of the advancements and challenges in scalable modular antenna arrays for 5G Massive MIMO (Multiple Input Multiple Output) networks, a pivotal technology for the next-generation wireless communication. Tracing the evolution of wireless technologies from 1G to 4G, the paper contextualizes the paradigm shift brought by 5G, characterized by enhanced spectral efficiency, massive device connectivity, and higher frequency bands. Focusing on Massive MIMO, the paper explores its role in augmenting network capacity and signal quality via advanced techniques like beamforming and spatial multiplexing. It delves into the intricacies of designing and implementing scalable modular antenna arrays, essential for the flexibility and optimization of rapidly evolving 5G networks. The review also covers mathematical modeling, advantages of Massive MIMO, and the challenges in integration, performance under diverse conditions, and cost-complexity balance. Concluding with insights into the commercialization trajectory of MIMO technology and its integration into modern telecommunications, the paper highlights the ongoing research directions and future potential of scalable modular antenna arrays in meeting the ambitious demands of 5G and beyond.

INDEX TERMS 5G, massive MIMO, modular antenna, wireless communication.

I. INTRODUCTION

The world of wireless communication has undergone a transformative journey, evolving through several generations of technologies, each marking a significant leap in capability, efficiency, and application. This evolutionary path has led us to the brink of the 5G era, a revolutionary stage in wireless communication that promises unprecedented speeds, reliability, and connectivity. The advent of 5G

The associate editor coordinating the review of this manuscript and approving it for publication was Hussein Attia $^{(D)}$.

technology is poised to redefine the landscape of digital communication and catalyze the growth of a plethora of industries and services [1]. Wireless communication has its roots in the late 19th century, with the first generation (1G) emerging in the 1980s, primarily facilitating analog voice communication. The subsequent generation, 2G, introduced digital voice services and basic data communication, laying the groundwork for more advanced applications. With 3G, the world witnessed a significant shift, enabling faster data transfer rates and better voice quality, thereby facilitating the early stages of mobile internet access and video calling.

The 4G era brought with it high-speed mobile internet, supporting video streaming, gaming, and advanced services, which laid the foundation for today's digital lifestyle [2].

The transition to 5G marks a paradigm shift in wireless communication. Unlike its predecessors, 5G is not just an enhancement of data rates but a comprehensive framework designed to support a wide range of applications including the Internet of Things (IoT), autonomous vehicles, telemedicine, and more. This transition is characterized by a move towards higher frequency bands, enhanced spectral efficiency, lower latency, and the ability to connect a massive number of devices concurrently [3]. The significance of 5G cannot be overstated. It is set to become the backbone of future digital ecosystems, driving economic growth, and fostering innovations across various sectors [4]. The high data rates, low latency, and massive device connectivity that 5G offers are critical for realizing the full potential of IoT, smart cities, virtual reality (VR), augmented reality (AR), and much more. Industries such as healthcare, automotive, entertainment, and manufacturing stand to benefit immensely from the enhanced capabilities of 5G networks. Additionally, 5G is expected to play a pivotal role in addressing some of the most pressing challenges of our times, including environmental sustainability, urbanization, and the digital divide. By enabling more efficient resource utilization, smarter infrastructure management, and broader connectivity, 5G could contribute significantly to sustainable development goals.

A key technology that underpins the 5G architecture is Massive Multiple Input Multiple Output (MIMO). Massive MIMO involves the use of a large number of antennas at the base station to simultaneously serve many users. This technology significantly increases the capacity of a wireless network without requiring additional spectrum, making it a cornerstone for achieving the high data rates and low latency promised by 5G [5]. The deployment of Massive MIMO systems is instrumental in overcoming the physical limitations associated with higher frequency bands used in 5G, such as path loss and attenuation. By exploiting techniques like beamforming and spatial multiplexing, Massive MIMO can enhance signal quality and system capacity, making it a crucial component for the success of 5G networks [6].

This review paper focuses on the advancements and challenges in scalable modular antenna arrays for Massive MIMO in 5G networks. The scalability of antenna arrays is critical for 5G deployment, as it allows for flexibility in network design and optimization according to varying user demands and environmental conditions. Modular antenna arrays, in particular, offer the advantage of easy upgradability and maintenance, which is essential in the rapidly evolving 5G landscape. This paper makes several significant contributions to the field of scalable modular antenna arrays for 5G Massive MIMO networks:

• Comprehensive Review: Provides a detailed review of the advancements and challenges in the design and implementation of scalable modular antenna arrays, encapsulating the evolution from 1G to 5G technologies and the paradigm shift introduced by 5G.

- Technical Insights: Offers in-depth technical insights into Massive MIMO technology, focusing on its role in enhancing network capacity and signal quality through advanced techniques like beamforming and spatial multiplexing.
- Design Considerations: Discusses critical design considerations for scalable modular antenna arrays, emphasizing the importance of flexibility and optimization in rapidly evolving 5G networks.
- Mathematical Modeling: Presents mathematical models to elucidate the functioning of Massive MIMO systems, including the base station architecture, beamforming techniques, and the impact on spectral and energy efficiency.
- Challenges and Solutions: Identifies key challenges in integrating scalable modular antenna arrays into 5G networks, such as integration complexities, spatial constraints, and pilot contamination, and proposes potential solutions and future research directions.
- Commercial and Scholarly Review: Analyzes the commercialization trajectory of MIMO technology and provides a scholarly review of recent research, highlighting the ongoing efforts and future potential of scalable modular antenna arrays in 5G and beyond.
- Future Directions: Outlines future research directions, emphasizing the need for advanced signal processing algorithms, dynamic beamforming techniques, and enhanced CSI acquisition methods to overcome current challenges and optimize 5G network performance.

This paper is organized as follows: Section II traces the evolution of wireless communication technologies from 1G to 4G, setting the stage for the transformative potential of 5G. Section III introduces the concept of Massive MIMO technology, its role in 5G networks, and the advantages it brings to wireless communication. In Section IV, we delve into the architecture of Massive MIMO systems, including mathematical modeling and the intricacies of beamforming and spatial multiplexing. Section V discusses the advancements in scalable modular antenna array designs, emphasizing their critical role in the adaptability and efficiency of 5G networks. Section VI presents the methodology for the literature survey along with a comprehensive scholarly review, analyzing recent research developments and their implications for the future of 5G and beyond. This section also addresses the commercialization aspects of MIMO technology, highlighting its journey from inception to its current pivotal role in 5G deployment. The paper concludes in Section VII by outlining the significant challenges faced in integrating scalable modular antenna arrays into 5G networks, proposing potential solutions, and suggesting directions for future research. This structure aims to provide a coherent and comprehensive overview of scalable modular antenna arrays in 5G Massive MIMO networks, from foundational concepts



FIGURE 1. 1G cellular architecture.

to future perspectives. Finally, section VIII concludes this paper.

II. EVOLUTION OF WIRELESS COMMUNICATION TECHNOLOGIES: FROM 1G TO 4G

The evolution of wireless communication, from the advent of 1G in the late 1970s to the revolutionary 4G era, has transformed mobile communication. Each generation, building on its predecessor, brought advancements like digital technology, mobile internet, and high-speed broadband.

A. 1G: THE BEGINNING OF MOBILE COMMUNICATION

The first generation of wireless cellular technology (1G) emerged in the late 1970s and early 1980s. This era marked the transition from traditional landline telephony to the first mobile telephones [7]. Fig. 1 depicts the basic architecture of an early mobile telecommunications network. At its core is the Mobile Telephone Switching Office (MTSO), which functions as the main control center, handling the routing and management of mobile phone calls. Each hexagon represents a cell site, part of a network of such sites that provide geographical coverage and facilitate wireless communication with mobile devices within their respective areas. The lines connecting the cell sites to the MTSO symbolize the transmission paths, which could be either wireline or microwave links. These paths enable the transfer of calls between the mobile network and the MTSO. On the other side of the MTSO are connections to the public switched telephone network (PSTN), depicted here as a local exchange, which in turn connects to landline telephones. This setup allows for seamless integration and communication between mobile users and traditional landline telephones, showcasing the interconnectivity of different communication infrastructures.

Key characteristics of 1G include: Analog Transmission: 1G networks were entirely analog, primarily used for voice communication. They utilized Frequency Division Multiple Access (FDMA) to allocate individual frequency bands to users.

Limited Capacity and Coverage: Due to analog technology, 1G had limited capacity and coverage. Calls often suffered from poor voice quality and were prone to eavesdropping.

First Mobile Phones: The introduction of 1G saw the first mobile phones, which were large, heavy, and had limited battery life.

The first generation primarily utilized omnidirectional antennas mounted on high towers to provide wide-area coverage [8]. These antennas were simple in design, focusing on maximizing coverage rather than capacity or data rates.



FIGURE 2. Architecture of a 2G cellular network.

B. 2G: DIGITAL REVOLUTION AND SMS INTRODUCTION

The second generation (2G) of wireless communication was introduced in the 1990s, marking a significant leap from analog to digital technology. Fig. 2 illustrates the architecture of a 2G cellular network, featuring key components that facilitate mobile communication. The mobile device, equipped with a SIM card, connects to the network via the Um interface to the Base Transceiver Station (BTS), which serves as the access point for wireless communication. Multiple BTSs are managed by the Base Station Controller (BSC), and together they form the Base Station Subsystem (BSS). The BSS is linked to the Mobile services Switching Center (MSC) via the A interface, which is the central component of the network's switching system responsible for routing calls and managing connections to other networks, like the Public Switched Telephone Network (PSTN). The MSC works in conjunction with databases such as the Home Location Register (HLR), Visitor Location Register (VLR), and the Authentication Center (AuC) for user authentication, registration, and to maintain subscriber information. The Gateway MSC (GMSC) is the interface that connects the cellular network to external networks, allowing for communication between different systems. The entire setup is designed to efficiently manage voice calls, SMS, and basic data services characteristic of 2G networks [9].

Key advancements in 2G include: Digital Signal Processing: 2G networks utilized digital signals, significantly improving voice quality, security, and capacity compared to 1G.

Introduction of SMS: 2G introduced Short Message Service (SMS), a revolutionary feature allowing text-based communication.

Data Services: Although limited, 2G began offering data services like email and basic internet, laying the groundwork for mobile data communication.

Global System for Mobile Communications (GSM): GSM became the standard for 2G, facilitating international roaming and compatibility.

With the advent of 2G and the transition to digital signals, more sophisticated antenna designs began to emerge. Sectorized antennas were introduced, dividing the cell site coverage area into sectors (typically three or six per cell site) to increase capacity and reduce interference [10]. This was a significant step towards more efficient use of the spectrum and improved service quality.



FIGURE 3. Architecture of a 3G cellular network.

C. 3G: THE DAWN OF MOBILE INTERNET

The third generation (3G) was launched in the early 2000s, bringing substantial improvements in data transmission and marking the beginning of the mobile internet era [9]. Fig. 3 outlines the architecture of a 3G cellular network, dividing the system into the Radio Access Network (RAN) and the Core Network, which consists of the Circuit Switched and Packet Switched domains. In the RAN, mobile devices (MN1, MN2) connect to the 3G Base Stations (BS1, BS2), which are then managed by the Radio Network Controller (RNC). The Core Network's Circuit Switched Domain handles voice calls and traditional SMS, with the Mobile Switching Centre (MSC) orchestrating the call setup, teardown, and routing to the PSTN. It is supported by the Visitor Location Register (VLR), Home Location Register (HLR), and Authentication Centre (AuC) for managing subscriber data and security. The Packet Switched Domain caters to data services, using the Serving GPRS Support Node (SGSN) and Gateway GPRS Support Node (GGSN) to manage data sessions and internet connectivity. The Gateway MSC (GMSC) provides the interface between the 3G network and external networks, enabling inter-network calling and connectivity [11]. This architecture enhances the capabilities of mobile networks, allowing for more advanced data services alongside traditional voice communication.

Key aspects of 3G include: Increased Data Rates: 3G networks offered significantly higher data rates, enabling functionalities such as mobile internet access, video calling, and multimedia services.

Wideband Code Division Multiple Access (WCDMA): This was a primary technology behind 3G, offering higher capacity and spectral efficiency.

Enhanced Security and Quality of Service (QoS): 3G networks provided improved security measures and better QoS, catering to more sophisticated mobile services.

Smartphones and Applications: The era of 3G saw the rise of smartphones and mobile applications, drastically changing the way users interact with mobile devices.

The third generation advanced antenna technology by implementing adaptive antenna systems (AAS) and the early forms of MIMO (Multiple Input Multiple Output) technology [12]. These developments allowed for better signal quality, higher data rates, and more efficient spectrum use, paving the way for the mobile internet era.



FIGURE 4. Architecture of an LTE (Long Term Evolution) network.

D. 4G: HIGH-SPEED MOBILE BROADBAND

Introduced in the late 2000s, the fourth generation (4G) further revolutionized mobile internet services [13]. Figure 4 illustrates the architecture of an LTE (Long Term Evolution) network, separating the control plane (responsible for signaling and control messages) from the user plane (handling data traffic). The User Equipment (UE), such as a smartphone, connects to the network via the eNodeB (evolved Node B), the base station of the LTE network. The control plane's signaling between the eNodeB and the core network occurs through the MME (Mobility Management Entity), which manages session states and user identity. The MME communicates with the HSS (Home Subscriber Server) via the S6a interface for authentication and subscription data. The user plane data flows from the eNodeB to the Serving Gateway (SGW) and then to the Packet Data Network Gateway (PGW), which provides connectivity with external networks like the Public Data Network (PDN) via the SGi interface. The SGW and PGW are linked through the S5 interface, allowing for user data transfer and mobility management [14]. This architecture supports efficient management of mobile broadband data and seamless connectivity for users as they move across the network.

4G's key features include: High-Speed Data Transmission: 4G networks, based on Long-Term Evolution (LTE) technology, offered unprecedented data speeds, enabling high-definition (HD) video streaming, advanced gaming, and high-speed mobile web access.

IP-Based Communication: 4G networks are entirely IPbased, facilitating seamless integration of voice, data, and multimedia services.

Improved Network Architectures: 4G introduced advanced network architectures, including evolved packet cores and IP multimedia subsystems, enhancing efficiency and scalability.

Proliferation of Mobile Applications: With the capabilities of 4G, the mobile app industry flourished, leading to an explosion of applications in every domain, from entertainment to finance.

The journey from 1G to 4G has been characterized by continuous innovation and technological breakthroughs. Each generation has built upon the successes and lessons of its predecessors, leading to more advanced, efficient, and user-friendly wireless communication technologies. This evolution has not only transformed the telecommunications industry but also significantly impacted the broader technological landscape, influencing everything from social interactions to business operations. As we transition into the 5G era, the groundwork laid by the previous generations is evident. The advent of 5G promises to leverage this legacy to offer even faster speeds, lower latency, and more reliable and ubiquitous connectivity, setting the stage for the next revolution in wireless technology.

With 4G and LTE, antenna technology saw the widespread adoption of MIMO, where multiple antennas were used at both the transmitter and receiver ends to significantly boost network capacity and speed. This generation also saw the introduction of beamforming techniques, enabling more targeted signal transmission and further enhancements in network efficiency and performance [15].

III. EMERGENCE OF 5G: OVERCOMING 4G LIMITATIONS AND PAVING THE WAY FOR FUTURE CONNECTIVITY

As the capabilities of 4G networks began to unfold, they set a new standard in mobile communication, offering high-speed data transmission and supporting a wide range of applications and services. However, as the technological landscape continued to evolve, certain limitations of 4G became apparent, necessitating the development of the next generation -5G.

A. LIMITATIONS OF 4G

Capacity and Network Congestion: With the exponential increase in the number of connected devices, 4G networks began to face challenges in handling massive device connectivity, leading to network congestion, especially in densely populated areas.

Latency Issues: While 4G networks offered reduced latency compared to previous generations, they still could not meet the requirements of emerging applications such as autonomous vehicles, real-time gaming, and certain industrial automation processes.

Energy Efficiency and Cost: The architecture and technology behind 4G are less energy-efficient, leading to higher operational costs. This is increasingly important as the world moves towards sustainable technology solutions.

Uniform Quality of Service (QoS): 4G networks struggle to provide uniform QoS across diverse geographical locations, especially in rural and remote areas.

Limited Scope for New Services: While 4G significantly improved data services, it had limitations in terms of supporting new technologies like the Internet of Things (IoT), augmented reality (AR), and virtual reality (VR) at scale.

B. DRIVERS FOR 5G DEVELOPMENT

The advent of 5G is not just an incremental improvement over 4G but a comprehensive overhaul designed to address these limitations and cater to the needs of an increasingly connected and digitalized world [16]. Fig. 5 showcases the network architecture for a 5G system, differentiating between the control plane and user plane. The control plane is responsible



FIGURE 5. Architecture of 5G network.

for signaling and control functions it includes various network functions (NFs) like the Network Slice Selection Function (NSSF), Network Exposure Function (NEF), Network Repository Function (NRF), Policy Control Function (PCF), Unified Data Management (UDM), Application Function (AF), Network Slice Selection Assistance Information (NSSAI), Authentication Server Function (AUSF), Access and Mobility Management Function (AMF), Session Management Function (SMF), and Service Capability Exposure Function (SCP). These NFs communicate with each other over specified service-based interfaces (such as Nnssf, Nnef, Nnrf, Npcf, Nudm, and Naf). The user plane, which handles data traffic, consists of the User Equipment (UE), Radio Access Network (RAN), and the User Plane Function (UPF). The UE connects to the RAN (which facilitates the wireless communication) over the N1 interface. The RAN connects to the control plane NFs through the N2 interface and to the UPF through the N3 interface. The UPF connects to the data network (DN) through the N6 interface, enabling user data to be transmitted to and from the internet or other data networks. The SMF, part of the control plane, manages sessions and interacts with the UPF via the N4 interface, orchestrating the flow of user data. This architecture supports the flexible and efficient management of network resources, user mobility, and data services in 5G networks [17].

Handling Massive Connectivity: One of the primary drivers for 5G is the need to efficiently handle the growing number of connected devices and the data traffic they generate, especially with the proliferation of IoT.

Ultra-Low Latency: 5G aims to drastically reduce latency to enable real-time communication and control, which is essential for applications like remote surgeries, autonomous driving, and advanced gaming.

Enhanced Data Rates: To support bandwidth-intensive applications like 4K/8K video streaming, virtual and augmented reality, and cloud-based services, 5G promises significantly higher data rates.

Energy Efficiency: 5G introduces more energy-efficient technologies and network architectures, reducing the environmental impact and operational costs.

Improved Coverage and Reliability: 5G is designed to provide more consistent and reliable coverage, including in challenging environments and remote areas.

Facilitating Advanced Technologies: 5G is seen as a key enabler for emerging technologies and innovative

applications, bridging the gap between potential and practical implementation.

Economic and Societal Benefits: Beyond technological advancements, the development of 5G is driven by its potential to generate significant economic growth, foster new industries, and contribute to societal advancements.

The role of antenna technologies becomes even more pivotal. 5G networks leverage advanced MIMO configurations, known as Massive MIMO, alongside dynamic beamforming and phased array technologies. These developments represent a quantum leap from the antenna technologies used in previous generations, offering unprecedented improvements in network capacity, coverage, and user experience [18]. The emergence of 5G is a response to the limitations of 4G and the growing demands of a digitally driven society. It represents a significant leap forward, not just in terms of speed and efficiency, but in enabling a new era of interconnectedness and innovation. As we step into the 5G era, it is poised to transform not only the telecommunications sector but numerous aspects of everyday life and business, heralding an era of unprecedented connectivity and technological possibilities.

IV. MASSIVE MIMO TECHNOLOGY IN 5G

Multiple Input Multiple Output (MIMO) is a technology that has become synonymous with modern wireless communication systems. At its core, MIMO leverages multiple antennas at both the transmitter and receiver to improve communication performance. By sending and receiving multiple data signals simultaneously over the same radio channel, MIMO systems effectively increase the capacity and reliability of wireless networks without requiring additional spectrum [19].

The initial applications of MIMO were relatively modest, focusing on enhancing the quality of wireless local area networks (WLANs). The IEEE 802.11n standard, for example, was one of the first to incorporate MIMO, leading to significant improvements in data throughput and signal range. In cellular networks, MIMO began to gain traction with the adoption of 4G LTE, where it helped to meet the growing demand for high-speed mobile data services.

The transition from MIMO to Massive MIMO has been driven by the relentless demand for higher data rates and the exponential increase in the number of connected devices. Massive MIMO extends the principles of MIMO by using a very large number of antennas (typically, tens or even hundreds) at the base station. This not only multiplies the capacity of a base station by orders of magnitude but also improves the transmission and reception of signals through techniques such as beamforming [20].

Massive MIMO is particularly significant in the context of 5G for several reasons. Firstly, it is one of the key enablers of the enhanced Mobile Broadband (eMBB) use case of 5G, which requires high data rates across a wide coverage area. Secondly, it facilitates a more efficient use of the spectrum, which is becoming increasingly scarce. Thirdly, Massive

MIMO's ability to direct energy in narrow beams helps in reducing interference and increasing energy efficiency, which is crucial for another 5G use case: the Internet of Things (IoT) [21].

The significance of Massive MIMO in 5G cannot be overstated. By significantly increasing the number of spatial channels, Massive MIMO allows 5G networks to serve many users simultaneously, with high data rates and over large distances. Furthermore, it is instrumental in realizing the full potential of millimeter-wave (mmWave) frequencies, which are a cornerstone of 5G's high-speed data capabilities [22].

The benefits of Massive MIMO in 5G extend beyond capacity and include substantial improvements in spectral efficiency, which measures how effectively a given bandwidth is utilized. This is achieved through more precise beamforming, which also contributes to energy efficiency by reducing wasted power.

A. ARCHITECTURE OF MASSIVE MULTIPLE INPUT MULTIPLE OUTPUT

Massive MIMO (Multiple Input Multiple Output) is a fundamental technology in modern wireless communication systems, particularly in 5G networks. Its structure and operation are designed to significantly enhance the capacity and efficiency of wireless networks [23]. Fig 6 illustrates the concept of Massive MIMO, a key technology in 5G networks designed to enhance capacity and efficiency through a structure that supports numerous simultaneous connections. In this system, a base station equipped with a large array of antennas (potentially in the hundreds) communicates with multiple user devices, each with typically one antenna. Through spatial multiplexing, the base station transmits distinct streams of data to different users over the same frequency channel, with the transmitted signal vector x being the product of the signal s and the beamforming matrix W. The user devices receive a combination of these signals plus noise. Mathematically, the received signal at each user device is a function of the channel response h and includes terms for both the intended signal and the interference from other users' signals, along with noise. The deployment of Massive MIMO thus significantly increases network capacity and efficiency, as it allows the simultaneous servicing of multiple users while maintaining high data rates and reducing interference, a cornerstone in delivering the vast improvements in bandwidth and latency promised by 5G technology.

B. MATHEMATICAL MODELING

Massive MIMO Base Station: The base station is equipped with an array of N antennas (where N can be in the order of hundreds), enabling it to handle numerous user devices (denoted as User 1 to User K) concurrently. Each user device is typically equipped with a single antenna.

Spatial Multiplexing: Spatial multiplexing is a technique that allows the transmission of multiple independent data



FIGURE 6. Massive MIMO architecture.

streams in the same frequency band, exploiting the spatial dimension. Mathematically, if s is a vector containing the signals for each user, the transmitted signals by the base station can be represented as:

$$X = Hs. \tag{1}$$

where X is the transmitted signal vector, and H is the channel matrix.

The channel matrix is a complex matrix representing the propagation from the base station antennas to each user's antenna, capturing the effects of fading and path loss [24], [25].

Beamforming: Beamforming is used to focus the transmission power towards specific users to improve signal quality and reduce interference. This is achieved by adjusting the phase and amplitude of the signal at each antenna. The vector wk represents the beamforming weights for user k, and the signal intended for that user is:

$$w_k^H s_k. (2)$$

where Sk is the signal for user k, and H denotes the Hermitian transpose [26], [27].

Channel State Information (CSI): The performance of Massive MIMO relies heavily on accurate CSI, which is represented by the matrix H. In TDD systems, the uplink and downlink channels are reciprocal, allowing the base station to estimate the downlink channel state from the uplink transmission, which is a major advantage in terms of overhead [28].

Mathematical Model: Consider K single-antenna user devices and a base station with N antennas. The base station transmits signals x such that:

$$x = \sum_{k=1}^{K} W_k s_k \tag{3}$$

where Sk is the data symbol intended for the k-th user, and Wk is the corresponding beamforming vector. The signal received by the k-th user is:

$$y_k = h_k^H x + n_k \tag{4}$$

where hk is the channel vector from the base station to the k-th user, and nk is the noise. The capacity of the channel to user k can be approximately given by:

$$C_{k} = \log_{2} \left(1 + \frac{|h_{k}^{H}x + w_{k}|^{2}}{\sum_{j \neq k}^{K} |h_{k}^{H}x + w_{j}|^{2} + \sigma^{2}} \right)$$
(5)

where σ^2 represents the noise power. This equation highlights the importance of designing the beamforming vectors Wk to maximize the signal-to-interference-plus-noise ratio (SINR) for each user [29], [30].

C. ADVANTAGES OF MASSIVE MIMO

The importance of Massive MIMO is underscored by its ability to handle numerous users simultaneously, making it a pivotal innovation for the ever-expanding wireless ecosystem [31].

1) INCREASED NETWORK CAPACITY

At the heart of Massive MIMO's benefits is its ability to vastly increase the network capacity. Traditional MIMO systems are limited by the number of antennas they can employ, but Massive MIMO breaks this barrier by utilizing potentially hundreds of antennas, which work together to form beams of focused energy. This architecture enables the base station to serve multiple users in the same time-frequency resource via spatial multiplexing, significantly enhancing the network's throughput [32].

2) SPECTRAL EFFICIENCY

Spectral efficiency is a measure of how efficiently a system uses its frequency spectrum. Massive MIMO markedly improves this efficiency through its large antenna arrays. Mathematically, the spectral efficiency (in bits/s/Hz) of a Massive MIMO system can be given by:

$$log_2(1 + SINR)$$
 (6)

where SINR stands for the signal-to-interference-plus-noise ratio. Due to the large number of antennas, the SINR can be substantially improved, thus increasing the spectral efficiency. Empirical studies, such as those presented at various IEEE conferences, provide real-world validation of these theoretical models, showing how Massive MIMO can exploit the available spectrum more effectively than ever before [33].

3) ENERGY EFFICIENCY

Energy efficiency in Massive MIMO systems is achieved through precise beamforming techniques that direct the transmission power specifically towards the intended users rather than dispersing it in all directions. This targeted approach not only improves the quality of the received signal but also significantly reduces the power wastage. Research indicates that Massive MIMO systems can be designed to be power-efficient, with studies highlighting their potential to reduce energy consumption per bit of transmitted data, thereby contributing to the sustainability goals of modern wireless networks [34].

4) IMPROVED USER EXPERIENCE

Massive MIMO enhances the user experience by providing high data rates and reducing latency, which are essential for the responsiveness and reliability of modern mobile services. Throughput is substantially increased, and the connection stability is improved, even at the cell edges where users typically experience lower data rates. Case studies, such as those documented by mobile operators during field trials, have reported notable improvements in QoS and QoE, confirming the user-centric benefits of deploying Massive MIMO technology [35].

5) INTERFERENCE REDUCTION

One of the traditional challenges in densely populated network environments is interference. Massive MIMO addresses this issue through smart antenna technology that can nullify interference signals, allowing for cleaner transmission channels. Simulation results and practical field tests have demonstrated the capability of Massive MIMO systems to distinguish between signals from different users effectively, even in highly congested scenarios, thereby mitigating the adverse effects of interference [36].

6) ROBUSTNESS IN HIGH MOBILITY

High mobility scenarios, such as those experienced by users on trains or in vehicles, demand robust and reliable communication links. Massive MIMO systems maintain their performance by rapidly adapting to changing channel conditions, a feature that is critical in high-speed scenarios. Published transport studies highlight the ability of Massive MIMO to provide stable connectivity and high data rates despite the high Doppler shifts and rapid channel variations associated with high mobility [37]. List of the key mechanisms that contribute to the robustness of Massive MIMO systems in high mobility scenarios include: Advanced Channel Prediction, Adaptive Beamforming, Diversity Gain, Fast Link Adaptation, Time-Frequency Resource Allocation.

7) FACILITATION OF ADVANCED TECHNOLOGIES

Massive MIMO is not only about enhancing current technologies but also about enabling new ones. It provides the necessary infrastructure to support the proliferation of IoT devices, the immersive experiences of AR and VR, and the complex requirements of smart cities and industrial automation. Industry reports have detailed the role of Massive MIMO in such advancements, emphasizing its contribution to the seamless integration of a multitude of devices and applications within the 5G and beyond ecosystem [38].

8) CHANNEL HARDENING

In Massive MIMO, due to the law of large numbers, the channel between the base station and the users tends to

become more deterministic as the number of antennas grows. This phenomenon, known as channel hardening, makes the wireless channel more reliable and predictable, which simplifies the signal processing and enhances the consistency of data rates experienced by the end-users [39].

9) REDUCED PILOT CONTAMINATION

Pilot signals are used in time- division duplexing (TDD) systems for channel estimation. In massive MIMO systems, with careful pilot assignment and due to the high spatial resolution, the impact of pilot contamination (where signals from users in other cells interfere with the pilot signals) can be mitigated. This is crucial for accurate channel state information (CSI), which is essential for optimal beamforming and scheduling [35].

Table 1 compares traditional MIMO and Massive MIMO systems, enhancing the details to provide a more comprehensive view.

V. SCALABLE MODULAR ANTENNA ARRAY DESIGNS

In the dynamic landscape of 5G networks, scalability is a critical component, primarily due to the diverse and evolving demands of applications ranging from high-speed mobile internet to massive IoT deployments [40]. Scalable antenna array designs are paramount in addressing these requirements, as they allow network operators to adaptively expand or modify their infrastructure to meet the current network demands without complete overhauls [41].

Scalable designs offer the flexibility to adjust the number of active elements in an antenna array, enabling operators to incrementally enhance the network capacity and coverage based on user demand and service uptake. This modularity is essential in managing the initial investment costs, as it prevents the need for premature infrastructure expansion and allows for a more organic growth strategy aligned with customer base expansion and technological advancements [42].

Moreover, the scalability in antenna designs is vital for ensuring that the networks are future-proof. As 5G technology continues to evolve and as we transition to 6G and beyond, having a scalable system means that networks can support higher frequencies and wider bandwidths, which are expected to be part of future standards [43]. This adaptability also aids in spectrum management, allowing operators to efficiently utilize their allocated frequencies and dynamically adjust to new ones as they become available.

In practical terms, scalability means that individual antenna elements or modules can be added to existing arrays without significant downtime or disruption to ongoing services. This approach is particularly beneficial in dense urban areas, where the demand for data can grow exponentially in a short period [44]. It also has implications for rural deployments, where the initial user density might not justify a massive deployment, but over time, as adoption grows, the network can be scaled to improve coverage and capacity.

Feature	MIMO	Massive MIMO	Notes
Antenna Quantity	Typically, 2-8	Often exceeds 64	Massive MIMO employs significantly more antennas, allowing
			for spatial multiplexing of a larger number of user signals.
Pilot Contamination Im-	Minimal	Elevated	The increased number of antennas in Massive MIMO can lead
pact			to higher pilot contamination, necessitating sophisticated pilot
			allocation strategies.
System Throughput	Moderate	Substantially Enhanced	Massive MIMO leverages its larger antenna arrays to boost
			throughput dramatically.
Antenna Coupling	Negligible	Noticeable	In Massive MIMO, the proximity of antennas can lead to
			coupling, which requires careful antenna design to mitigate.
Bit Error Rate (BER)	Comparatively Higher	Reduced	The ability of Massive MIMO to focus energy more precisely
			leads to improved signal quality and lower BER.
Noise Immunity	Average	Superior	Massive MIMO systems are more resistant to noise due to
			increased diversity and signal processing capabilities.
Diversity/Capacity Gains	Moderate	Exceptionally High	Massive MIMO significantly enhances capacity gains due to
			the ability to serve multiple users concurrently in the same
	~		frequency band.
Energy Efficiency	Conventional	Optimized	Despite its complexity, Massive MIMO can be more energy-
			efficient due to focused beamforming that reduces wasted
	T	TT' 1	power.
Infrastructure Cost	Lower	Higher	The initial cost of Massive MIMO is higher due to more so-
			phisticated hardware, but this can be offset by efficiency gains
Openational Complexity	Stondord	Complay	and the admity to serve more users.
Operational Complexity	Standard	Complex	Massive MIMO introduces additional complexity in signal
Saalability Potential	Limited	Extensive	Massive MIMO offers a more scalable solution, and interference management.
Scalability Fotential	Linned	Extensive	wassive windo offers a more scarable solution, enabling fiet-
			tively
I ink Stability	Stable Under Certain Con-	Robust in Various Scenar-	Massive MIMO provides more stable links by exploiting large
Link Stability	ditions	ios	antenna arrays, even in challenging environments
Antenna Correlation	Less Concerning	Managed Carefully	In Massive MIMO, due to the dense arrangement of antennas
			correlation can affect performance if not properly managed
Deployment Flexibility	Rigid	Adaptable	The modular nature of Massive MIMO arrays allows for a more
	6	1	flexible deployment to meet diverse service requirements.
Maintenance and	Standardized	Modular	Massive MIMO systems facilitate easier upgrades and main-
Upgrades			tenance due to their modular design, allowing for individual
			components to be replaced or upgraded as needed.
User Capacity	Fixed	Dynamically Scalable	Massive MIMO can dynamically adjust to user capacity de-
			mands, making it ideal for fluctuating network loads.
Spectrum Utilization	Efficient	Highly Efficient	With advanced algorithms, Massive MIMO optimizes spec-
			trum use, leading to improved overall network performance.
Technological Evolution	Established	Progressively Advancing	Massive MIMO technology continues to evolve, incorporating
			advancements in AI and machine learning for enhanced perfor-
			mance.

TABLE 1. Comparative analysis of MIMO and massive MIMO architectures in wireless communications.

Finally, the importance of scalability is not limited to capacity and coverage. It also encompasses the ability to integrate new technologies, such as beamforming and full-dimension MIMO, which require sophisticated and flexible antenna designs. Scalable arrays can support these technologies by enabling the precise control of a large number of antenna elements, which is crucial for optimizing the network's performance and delivering the multi-gigabit speeds promised by 5G [45]. In essence, scalable modular antenna array designs are the backbone of a flexible, cost-effective, and future-ready 5G network infrastructure. They are the key to meeting the high demands of today's mobile users and tomorrow's technological challenges, making them an indispensable aspect of modern wireless communication systems [46].

The modular design approach in antenna arrays is a sophisticated concept that aligns with the modern requirements of flexibility and scalability in wireless network infrastructure, particularly in the context of 5G and beyond. A modular antenna array is composed of multiple sub-array modules that can be individually controlled and combined to form a larger, cohesive antenna system. Fig. 7 illustrates a detailed schematic of a modular antenna array design, specifically tailored for scalable implementation in advanced wireless communication systems like those used in 5G [47]. This design epitomizes the concept of modularity, where each component can be independently optimized, replaced, or upgraded.

Tile Antenna Subarray: The fundamental building block of this modular design is the 'Tile Antenna Subarray', which consists of multiple radiating elements (the orange rectangles) that form a subarray. These tiles are critical to the array's modularity, allowing for a scalable and customizable approach to constructing the antenna system [48].

Beamforming Integrated Circuit (BFIC): The BFIC located at the back of each tile handles the complex



FIGURE 7. multi-tile ("modular") antenna array design.

beamforming tasks. Beamforming is essential for directing the antenna's power towards specific users and away from others, thereby improving signal quality and reducing interference. The BFIC is instrumental in dynamically controlling the phase and amplitude of the signal fed to each radiating element [49].

RF Feeding Network: The RF feeding network (highlighted in yellow) is responsible for distributing the RF signals to each tile. This network ensures that the signal is correctly split and delivered to each subarray, maintaining the necessary phase and amplitude characteristics to achieve the desired beamforming effect [48].

Microstrip to Microstrip Transition: The transition components (shown in blue) are crucial for maintaining signal integrity when moving from one type of transmission line to another. In this context, they ensure efficient signal transfer between different sections of the RF feeding network, which is often a challenge in high-frequency designs [50].

DC/COM/Signaling Network: This layer provides the necessary control and communication links between the tiles and the central processing unit of the antenna system. It includes power distribution (DC), control signaling for beam steering and beam shaping (COM), and other necessary signaling pathways that allow the array to operate as a cohesive system [48].

Flexible Tiling Layer: The flexible tiling layer indicates the potential for rearranging or adding tiles to the system. This design feature enables the antenna array to physically expand or contract in size, accommodating different coverage areas or capacity requirements [47].

Tile Placement Slot: The tile placement slots are part of the mechanical structure that allows individual tiles to be added, removed, or replaced. This aspect of the design is crucial for the scalability of the system, ensuring that modifications to the antenna array can be made with minimal disruption to the overall operation [51]. The fundamental concepts associated with scalable modular antenna array designs are as follows;

Flexibility through Modularity: The essence of modularity in antenna design lies in the ability to compose, decompose, and reconfigure antenna elements to meet specific performance criteria. Each module functions as an independent unit with a defined set of radiating elements, electronics, and potentially its own signal processing capabilities. Modules can be added, removed, or rearranged based on the coverage requirements, signal environment, and capacity needs of the network [52].

Modularity in antenna design is represented mathematically by the ability to express the array factor as a sum of the contributions from each module. If $A(f, \theta, \phi)$ is the array factor for the entire array as a function of frequency f and angles θ and ϕ , then for n modules, it can be expressed as:

$$A(f,\theta,\phi) = \sum_{i=1}^{n} M_i(f,\theta,\phi)$$
(7)

where Mi is the contribution from the i-th module.

Rapid Deployment and Maintenance: Modular arrays facilitate rapid deployment and ease of maintenance. Because individual modules are self-contained, they can be quickly replaced or upgraded without affecting the entire array. This plug-and-play nature simplifies maintenance and reduces network downtime, which is critical in maintaining the high reliability standards of 5G networks [53].

The rapid deployment and maintenance can be framed in terms of the Mean Time to Repair (MTTR) which is a standard measure of maintainability. For a modular array, if the MTTR for a single module is t, the overall MTTR for the array is significantly reduced as only the affected module needs to be addressed:

$$MTTR_{array} \approx t$$
 (8)

This is in contrast to non-modular systems where the MTTR could scale with the size of the array.

Adaptive Performance: One of the key advantages of modular arrays is their inherent adaptability. They can dynamically adjust their operational parameters, such as radiation patterns, frequency bands, and polarizations, to adapt to changing network conditions. This adaptability is achieved through active control of the modules' signal paths and phase shifters, allowing the network to respond to varying user demands in real-time [54]. The adaptability of an antenna array can be quantified by the beam steering capability, which is a function of the phase shift $\Delta \phi$ applied by the phase shifters in each module. The steering angle θ_s can be calculated as:

$$\theta_s = \arcsin\left(\frac{\lambda\Delta\phi}{2\pi d}\right) \tag{9}$$

where λ is the wavelength of the signal, and *d* is the distance between adjacent elements in the module.

Cost-Effective Scaling: Modular designs are also costeffective. By allowing network operators to scale their systems incrementally, they can align their capital expenditures more closely with user growth and revenue. This staged deployment strategy is more economical compared to deploying a large-scale antenna system from the outset, which may remain underutilized for an extended period [55]. The economic scaling of modular designs can be related to a cost function C(N) which depends on the number of modules N. Ideally, this cost function has sub-linear growth, meaning:

$$C(N + \Delta N) < C(N) + C(\Delta N)$$
(10)

This reflects the reduced incremental cost as the system scales.

Enhanced System Reliability: In modular arrays, the failure of a single module does not incapacitate the entire array. Instead, the system can continue functioning, albeit with reduced capacity, which enhances the overall reliability of the network. This fault tolerance is particularly important in mission-critical applications where consistent service availability is non-negotiable [56]. System reliability R_{system} can be modeled as the product of the reliabilities of individual modules R_i . If a failure in one module does not lead to total system failure, the system reliability for *n* independent modules is:

$$R_{\text{system}} = 1 - \prod_{i=1}^{n} (1 - R_i)$$
(11)

This shows that even if individual modules have a certain probability of failure, the overall system can maintain a high level of reliability.

Design Considerations: In the design of modular antenna arrays, several factors are considered to ensure optimal performance. These include the inter-module spacing, which affects the array's grating lobes and side lobe levels, and the module's beam steering range, which determines the flexibility in directing the beam to different spatial regions. Additionally, the integration of advanced materials and miniaturized electronics within each module plays a critical role in the array's performance [57]. The performance of a modular antenna array can be quantified by considering the array's directivity D and side lobe level *SLL*. The directivity is influenced by the inter-module spacing d and can be described as:

$$D \approx \frac{4\pi A_{\rm eff}}{\lambda^2} \tag{12}$$

where A_{eff} is the effective aperture area of the array. The side lobe level is related to the array's layout and element pattern, and minimizing *SLL* is crucial for reducing interference. The design must strike a balance between directivity and side lobe levels to optimize performance.

A. STRUCTURES OF SCALABLE MODULAR ANTENNAS

The scalability and modularity of antenna designs are crucial for the dynamic nature of 5G networks. Various innovative antenna structures have been developed to meet these needs, each offering unique benefits:

 Phased Array Antennas: These antennas consist of multiple radiating elements whose signals are phaseshifted and combined to form directional beams. This electronic steering capability allows for rapid reconfiguration of the beam direction to support user

is achieved by adjusting the number of elements in the array, enabling the system to cater to varying coverage and capacity requirements without physical reorientation [58].Massive MIMO Arrays: Characterized by their use of

• Massive MIMO Arrays: Characterized by their use of a large number of antenna elements, Massive MIMO arrays can serve numerous users simultaneously within the same frequency band through spatial multiplexing. This significantly enhances network capacity and efficiency. The modular nature of these arrays allows for incremental expansion, providing a straightforward path for network growth as user demand increases [59].

mobility and manage interference effectively. Scalability

- Reconfigurable Intelligent Surfaces (RIS): RIS represent a paradigm shift in wireless communications by controlling the propagation environment itself. Comprising numerous small, programmable elements, an RIS can reflect, refract, or absorb incoming signals to enhance or extend coverage. This technology offers a novel form of scalability, where the size and configuration of the surface can be adapted to meet specific network enhancements [60].
- Metamaterial Antennas: Leveraging the unique properties of metamaterials, these antennas can achieve negative refraction and superlensing effects, enabling more compact and efficient designs. Metamaterial antennas can be engineered to specific frequency responses and radiation patterns, making them highly adaptable and scalable for various applications, from mobile devices to base stations [61].
- Active Antenna Systems (AAS): AAS integrate active components, such as amplifiers and digital signal processors, directly with the antenna elements. This integration supports more intelligent and adaptive beamforming capabilities, allowing for real-time network optimization. The modular design of AAS enables easy expansion of network capabilities by adding more active antenna units as needed [62].
- Fractal Antennas: Utilizing the recursive nature of fractal geometries, these antennas can operate over a wide range of frequencies and are inherently multi-band. Their self-similar design allows for scaling in size without significantly affecting performance, making them versatile for various applications, from handheld devices to large-scale base station deployments [63], [64].

VI. METHODOLOGY FOR LITERATURE REVIEW

The selection of research works for analysis in this paper followed a systematic approach to ensure relevance and comprehensiveness. The process involved several key steps:

• Initial Search: We began with an extensive search in prominent databases such as IEEE Xplore, ScienceDirect, ResearchGate, Wiley Online Library and Google Scholar, using keywords related to "5G," IEEE Access

- and "wireless communication technologies."
 Inclusion Criteria: Research works were selected based on their relevance to the core themes of this paper, including technological advancements in antenna arrays, Massive MIMO systems, and their application in 5G networks. Priority was given to articles published in the last five years to ensure the timeliness of the information. However, some seminal works and foundational studies, irrespective of their publication date, were also included. This was done to ensure a comprehensive understanding of the subject matter, acknowledging that certain older references continue to hold significant value and insight into the development and theoretical underpinnings of current technologies.
- Exclusion Criteria: Works that did not directly address the design, implementation, or challenges of scalable modular antenna arrays in the context of 5G networks were excluded. Additionally, we filtered out articles that were not peer-reviewed to maintain the quality and credibility of the sources cited.
- Quality Assessment: Each potential source underwent a quality assessment, evaluating the depth of analysis, methodological rigor, and contribution to the field. This ensured that only high-quality, impactful studies were included in our review.
- Cross-Referencing: To broaden the scope of our literature review, we also examined the references within the selected articles for additional relevant works that may have been missed in the initial search.
- Final Selection: The final set of included works represents a balanced mix of theoretical research, practical case studies, and scholarly reviews, providing a comprehensive perspective on the subject.

A. SCHOLARLY REVIEW

In recent years, the field of scalable modular antenna arrays for 5G Massive MIMO networks has witnessed a significant surge in scholarly interest, as evidenced by the increasing volume of literature on the subject. This section presents a comprehensive review of the advancements and challenges documented in academic and industry research, highlighting the prolific contributions and the evolving trends that are shaping the future of wireless communication technologies. The accompanying bar chart shown in Fig 8 distinctly illustrates the exponential growth in research output from 2013 to 2023, underlining the intensified focus on optimizing antenna array designs to meet the ambitious demands of 5G and beyond [65].

According to [3], they addressed the methodological advancements in Massive MIMO systems for 5G and beyond, focusing on the development of various algorithms to improve device detection and channel estimation. Hybrid Generalized Approximate Message Passing (GAMP) methods that combine nonlinear measurements and group sparsity



FIGURE 8. Yearly distribution of scholarly publications on scalable modular antenna arrays for 5G massive MIMO networks (2013-2023).

are used to enhance device detection performance. The paper also highlights the necessity of accurate Channel State Information (CSI) for effective massive access, where the CSI is typically acquired through channel estimation based on pilot sequences sent by devices. Innovative approaches like Joint Activity Detection and Channel Estimation (JADCE) algorithms have been developed to manage massive connectivity. The paper further discusses the results showing the potential of asymptotic regimes where the number of users and antennas grows infinitely, allowing for near-zero miss detection and false alarm probabilities in activity detection algorithms. Additionally, hybrid precoding techniques have been proposed to reduce the number of RF chains required while retaining the benefits of massive MIMO, thus addressing the high cost of Analog-to-Digital Converters (ADCs) which is a significant issue in the deployment of massive MIMO systems. The advantages of these advancements are clear: enhanced detection performance, more efficient use of RF chains, and the mitigation of ADC costs. However, there are limitations such as the high design complexity of hybrid precoding schemes and the need for further research to optimize JADCE algorithms and other areas critical to the deployment of massive MIMO in real-world environments. The paper underscores the importance of ongoing research to address these challenges, with a comprehensive overview serving as a foundation for future work in the field.

According to [66], an advanced algorithmic approach designed to refine network coverage and throughput was presented. The method begins by leveraging the technical nuances of 5G NSA architectures to extract precise geolocational coordinates. Building on this foundational data, the study unveils a refined Massive MIMO weight optimization technique informed by MR-derived longitudinal and latitudinal insights. An intelligent weight selection mechanism is employed, navigating the complex landscape of MIMO weight configurations to pinpoint the most effective solution. To address the constraints inherent in existing systems, the paper advances a streamlined variant of the algorithm. This adaptation ensures more practical application, enhancing the method's appeal and utility across broader network deployments.

Based on the research by [67], the synergy of Massive MIMO and heterogeneous networks are crucial to the evolution of 5G technologies. The study explores the deployment of Massive MIMO within macro-cells, complemented by an array of small-cells across various tiers, aiming to amplify network reach and capacity. A key focus is the investigation of the impacts that a limited quota has on K-tier heterogeneous networks, particularly under conditions of Rayleigh fading channels. The paper adopts a novel approach to user admission, akin to a college admission process, to analyze network dynamics. It presents closedform solutions for critical metrics such as user association probability, coverage probability, and average data rate for both macro and small base stations. The numerical outcomes of the research underscore the substantial influence that quota limitations exert on system performance, offering valuable insights for the optimization of 5G network infrastructures.

According to [68], the development of a four-branch, reduced-complexity, load-modulated MIMO transmitter is presented, optimized for operation at 3 GHz. This innovative transmitter is capable of generating 64-QAM constellations directly at the antenna elements. The distinct feature of this design is its method of waveform generation, achieved by varying the impedance parameters of load modulator circuits linked to the antennas, while maintaining a constant output from the oscillator. This approach allows the use of a single RF chain to drive the entire transmitter, with the power amplifier tasked with amplifying a constant signal, thereby obviating the need for operation in the back-off region. The integration of a four-way Wilkinson power divider effectively distributes power across the transmitter's four branches. Microstrip patch antennas are employed in conjunction with the load modulators to emit the required signals. The inclusion of RF isolators is crucial, as they absorb reflections from each load modulator, ensuring that each branch can independently generate its desired constellation while keeping the system continuously matched at the input. The design and simulation of this transmitter were conducted using Advanced Design System (ADS), leading to a comprehensive characterization. Notably, this transmitter design eliminates the necessity for mixers and digital-toanalog converters (DACs), offering a simplified architecture.

In the study [69], the field trial outcomes for a massive multi-user MIMO (MU-MIMO) system operating at 28 GHz with a 500 MHz bandwidth are detailed. The trials employ a hybrid beamforming approach with block diagonal precoding, a method that relies on uplink sounding to gather channel state information, based on the principle of channel reciprocity in a time-division duplex system. The experiments focus on 8-user MU-MIMO scenarios featuring varied user distributions, assessing both downlink transmission performance and propagation channel characteristics. The findings from these trials reveal that the implementation of digital

precoding significantly enhances system performance, with the ability to achieve over 20 Gbps in total throughput across diverse user distribution scenarios. Furthermore, the efficacy of the precoding technique is specifically gauged by its success in mitigating inter-user interference, demonstrating its critical role in optimizing the performance of MU-MIMO systems. Table 2, Table 3 and Table 4 presents a comprehensive review summarizing key scholarly works in the field of Massive MIMO and related technologies, highlighting the distinct methodologies and operational mechanisms employed in each study:

B. COMMERCIALIZATION ASPECT

MIMO technology's journey began in 1996 with the revelation that colocated antennas could significantly amplify the capacity of wireless communication systems. This advancement was rooted in multidimensional signal processing and natural multipath propagation. A notable milestone was the development of MIMO OFDM fixed wireless links, which showcased impressive error-free operation over six miles using just one watt of transmitted power. This innovation not only paved the way for MIMO's integration into cellular wireless technologies but also set the stage for its adoption in LTE mobile networks and Wi-MAX [89].

The late 1990s and early 2000s witnessed the rise of MIMO technology, significantly shaped by companies like Airgo Networks, later acquired by Qualcomm in 2006 [90]. In 2004, Airgo Networks unveiled True MIMO, the first MIMO-enabled Wi-Fi device in the market, marking a significant advancement in wireless networking. This technology, utilizing spatial multiplexing, enhanced data speeds and reliability by broadcasting multiple data streams concurrently across a single frequency band.

The development of MIMO technology catalyzed the creation of antennas for 5G devices by several companies, aiming to achieve superior performance, capacity, and coverage. Notable contributors included Laird Connectivity, Kathrein, Rosenberger, CommScope, TESSCO, ZTE Corporation, PCTEL, Wilson Electronics, Taoglas, and Qorivo. The first commercial MIMO system was developed by Iospan Wireless Inc. in 2001. Following this, tech giants like Intel and Broadcom introduced novel communication techniques based on MIMO technology, significantly enhancing wireless LAN network performance [90].

The evolution of MIMO systems was driven by the objective to boost wireless communication performance, leveraging the spatial diversity and multipath propagation properties of radio waves. Specialized companies such as Shenzhen VLG Wireless Technology Co. Ltd. [91], Chinmore Industry Co., Ltd. [92], and REMO electronics emerged, focusing on the design and production of a variety of antennas, including those for MIMO applications [93].

Airgain, Inc. has become a leading supplier of advanced antenna systems, striving to provide a significant overthe-air (OTA) throughput boost over competing systems [94]. Collaborating with OEMs and ODMs, they aim to enhance the industrial IoT experience. Airspan Networks Inc., a prominent MIMO antenna manufacturer, designs a wide range of MIMO antennas for diverse wireless communication applications [95].

Huawei, a global leader in 5G antenna technology, offers an extensive array of multiband and smart antennas [96], while Taoglas Limited specializes in custom antenna designs for IoT applications [97]. Panorama Antennas Ltd., based in the UK, produces a variety of wireless antennas suitable for 2G to 5G frequencies [98]. Ericsson, a Swedish multinational company, supplies products and services for 5G networks, including massive MIMO antennas designed to enhance network capacity and coverage, supporting a broad range of frequencies and compatible with both 4G and 5G networks [99].

The commercial success of MIMO technology has been marked not only by its widespread adoption in modern telecommunications but also by its continuous evolution to meet the growing demands for high-speed data transmission and enhanced network capacity. This evolution represents a significant technological advancement in wireless communication, driven by both scientific discovery and commercial interests.

VII. CHALLANGES AND FUTURE DIRECTIONS

The rapid advancement and deployment of 5G Massive MIMO (Multiple Input Multiple Output) networks bring forth a set of unique challenges, particularly in the context of scalable modular antenna arrays. These challenges are pivotal in determining the efficiency, performance, and overall success of 5G networks. Below is a comprehensive overview of these challenges:

1. Integration and Compatibility: Integrating advanced antenna systems into existing network infrastructures poses significant challenges. This includes compatibility issues with legacy systems and the complexity of deploying largescale antenna arrays. Ensuring seamless integration while upgrading network capabilities is a balancing act that requires meticulous planning and execution.

2. Spatial and Electromagnetic Constraints: The physical integration of advanced MIMO antennas into varied network environments must consider spatial constraints and electromagnetic compatibility with other system components. The design and placement of antenna arrays must account for the limited space, especially in urban environments, and avoid interference with existing electronic and communication systems.

3. Performance under Diverse Conditions: Scalable modular antenna arrays must perform optimally across a wide range of conditions, including different frequency bands and user scenarios. This requires antennas to be versatile and adaptable to various environmental factors, user densities, and mobility patterns.

4. Cost and Complexity: The development of scalable modular antenna arrays involves high costs and complexity,

both in terms of design and deployment. Balancing costefficiency with advanced technological requirements is a significant challenge, especially when aiming for widespread implementation.

5. Pilot Contamination: In multi-cell TDD networks, non-orthogonal pilot schemes are a major source of pilot contamination. This issue, arising due to the limitations of coherence time, impacts the channel estimation process and, subsequently, the overall network performance. Developing effective mitigation strategies is crucial.

6. High-Speed Data Transmission: The goal of achieving high-speed data transmission with minimal path loss and consistent radiation patterns is a key challenge. The antenna design must ensure reduced delay and efficient data handling to meet the growing demand for high-capacity wireless communication.

7. Signal Processing Complexity: The increased number of antennas in Massive MIMO systems leads to higher signal processing complexity. Efficient algorithms for signal encoding, decoding, and interference mitigation are needed to manage this complexity without compromising system performance.

8. Beamforming Challenges: Advanced beamforming is critical in Massive MIMO for targeting signals towards specific users. The challenge lies in dynamically adjusting beam patterns in real-time to optimize network performance, especially in dense urban environments with high user mobility.

9. Channel State Information (CSI) Acquisition: Accurate and timely acquisition of CSI is vital in Massive MIMO systems. However, obtaining CSI in dynamic environments, where user location and channel conditions change rapidly, presents significant technical challenges.

10. Scalability vs. Performance Trade-off: Ensuring that the antenna array is scalable while maintaining or improving performance is a key technical challenge. This involves designing systems that can be expanded easily without significant increases in cost, complexity, or energy consumption.

11. Inter-Cell Interference: In dense network deployments, managing inter-cell interference becomes increasingly challenging. Techniques to mitigate this interference without causing additional overhead are crucial for maintaining network quality.

12. Hardware Limitations and Calibration: The physical limitations of antenna design, such as size, weight, and material constraints, pose challenges. Additionally, maintaining calibration across large arrays of antennas to ensure coherent operation is technically demanding.

13. Spectrum Utilization and Efficiency: Optimizing spectrum utilization while maintaining high energy efficiency is a technical hurdle. This involves developing antenna designs and signal processing techniques that maximize data throughput within the available spectrum.

14. Backhaul Capacity: The increased data rates in Massive MIMO systems require a backhaul network with high capacity. Ensuring that the backhaul can support the data

TABLE 2. Overview of research on massive MIMO and related technologies in 5G networks (part 1).

Citation	Method	Explanation	Advantages	Limitations
[70]	Software-Defined Network (SDN) in Massive MIMO HetNets	Utilizes SDN as the cen- tral controller for radio re- source management, focus- ing on managing spatial do- main information and MIMO coordination. The SDN con- troller collects user channel- state information, calculates null-space for victim users, and applies linear precoding	Offers high quality of ser- vice with lower computa- tion complexity; effective in managing complex MIMO coordination.	Potential scalability issues in very large network environ- ments and reliance on the efficiency and robustness of the SDN controller, which could be a bottleneck in dy- namic and high-demand sce- narios.
[71]	Multi-Objective Evolutionary Algorithms (MOEAs)	Employs MOEAs, particularly non-dominated sorting genetic algorithm- II and speed-constrained multi-objective particle swarm optimization, for optimizing massive MIMO integration in 5G. Includes a decision maker for selecting optimal solutions.	Demonstrates the effectiveness of MOEAs in addressing multi-objective optimization challenges in 5G.	The complexity of algorithms and their real- world application scalability might be challenging.
[72]	Massive MIMO Channel Modeling with Terminal Antenna Characteristics	Presents a simulation frame- work that incorporates ter- minal antenna characteristics for massive MIMO channel modeling, using Sony Xperia handsets at 3.7 GHz for eval- uation.	Provides close-to-real mea- sured performance; useful for antenna evaluation in pre- coded massive MIMO sys- tems.	Specific to certain hardware and frequency, which might limit general applicability.
[73]	L-shape Array Technique in Massive MIMO	Proposes an L-shape array- based technique to reduce cross-user correlation (CUC) in near line-of-sight massive MIMO scenarios without in- creasing signal processing complexity.	Effectively reduces CUC and improves sum-rates, performing close to Gaussian channels even in tightly coupled scenarios.	Requires additional compo- nents per RF chain at the base station, potentially in- creasing the system's hard- ware complexity.
[74]	Frequency Selective Surface (FSS) for Antenna Interfer- ence Mitigation	Introduces a method using FSS between antennas of dif- ferent frequencies to mini- mize interference. The FSS acts as a transparent structure for one antenna and as an artificial magnetic conductor for another.	Enhances individual antenna performance and maintains radiation patterns; innovative application in different an- tenna decoupling.	The focus is on specific frequency bands of 5G, which might limit its application across other frequency ranges.
[75]	Survey on Pilot Contamina- tion in Massive MIMO	Conducts an extensive sur- vey on pilot contamination in TDD Massive MIMO sys- tems, including its sources and impacts, and reviews mitigation strategies.	Offers comprehensive insights into the challenges and solutions for pilot contamination in Massive MIMO.	Focuses predominantly on pilot contamination, poten- tially overlooking other Mas- sive MIMO challenges.
[76]	Hybrid n-Bit Heuristic Assisted-VBLAST (HHAV) Algorithm	Introduces the HHAV algo- rithm to optimize decoding in massive MIMO systems, tested in dynamic Rayleigh fading channels. The study also evaluates the AMP al- gorithm with ternary and Gaussian distribution thresh- old functions.	Demonstrates significant performance improvements over existing detection systems in bit error rate and computational complexity.	The complexity of the algo- rithm might pose challenges in practical implementation.
[77]	Enhanced Geometry-Based Stochastic Channel Model	Proposes a new channel model for 5G M-MIMO that accounts for spherical wavefront and array nonstationary, using spatial geometric manipulations and a general Markov process.	Offers a more accurate repre- sentation of M-MIMO chan- nel characteristics in real en- vironments.	May require complex computations to accurately model real-world propagation environments.

TABLE 3. Overview of research on massive MIMO and related technologies in 5G networks (part 2).

Citation	Method	Explanation	Advantages	Limitations
[78]	12-Port Antenna Array for 5G MIMO	Details the design and test- ing of a 12-port antenna ar- ray suitable for various LTE bands used in 5G MIMO ap- plications, with different an- tenna element types for opti- mized performance.	Ensures wideband coverage and high efficiency with ex- cellent MIMO performance, even under hand phantom ef- fects.	Focused on specific LTE bands, which might limit its application in other fre- quency ranges.
[79]	RF Signal-Based Moving Direction Sensing Scheme in 5G Massive MIMO	The method estimates move- ment speed and direction in a 5G Massive MIMO system by analyzing RF signal en- ergy distribution and using antenna geometry and target locations.	Offers a high-accuracy alter- native to traditional orienta- tion sensors, with less than 1.5m/s speed estimation er- ror and within 2 degrees di- rection estimation error.	The effectiveness of the method may depend on specific scenarios and the precision of base station location data.
[80]	Structured Massive Access in Cell-Free Massive MIMO Systems	Proposes a new framework for scalable massive access in cell-free Massive MIMO, including an initial access algorithm, P-LSFD strategy, pilot assignment schemes, and fractional power control policy. Derives new closed- form SE expressions with MR combining.	Demonstrates high spectral efficiency and scalability, with the proposed components outperforming benchmarks and offering a balance between user fairness and average SE.	Specifics of implementation in varying network environments and potential complexity of the proposed strategies might be challenging.
[81]	Analysis of Evolving Mas- sive MIMO in 3GPP Re- leases	Analyzes massive MIMO's evolution in 3GPP Releases 15-17, focusing on CSI, beam management, and Release 18's challenges, with system simulations and field trials for validation	Provides in-depth insights into standardization progress and practical enhancements in massive MIMO, along with real-world performance data.	Focuses on specific aspects of 3GPP releases, which might not cover all emerging challenges in the broader 5G ecosystem.
[82]	Review of 5G Massive MIMO Antenna Technologies	Discusses the development and impact of 5G Massive MIMO antennas, focusing on mutual coupling reduction techniques. Highlights the advancements in antenna design for improving the performance of future-generation wireless systems.	Emphasizes the substantial improvements in transmis- sion speeds, reliability, and spectral efficiencies offered by 5G Massive MIMO an- tennas.	Not fully address the chal- lenges in integrating these advanced antenna systems into existing network infras- tructures, including compat- ibility issues and the com- plexity of deploying large- scale antenna arrays.
[83]	Integrated mMIMO Antenna System with Metamaterial	Proposes an integrated mas- sive MIMO antenna system loaded with metamaterial for 5G, featuring duple nega- tive characteristics and broad bandwidth. The design in- cludes eight subarrays with three-layer construction, op- erating at 3.5 GHz.	Offers high port isolation, significant gain improvement, and enhanced MIMO performance with ECC less than 0.0001 and overall efficiency above 90%.	The complexity of the an- tenna design and the spe- cific focus on the mid-band of 5G may limit its appli- cability across the entire 5G spectrum.
[84]	Comprehensive Review of MIMO Antenna Design for 5G	Presents a detailed review of MIMO antenna design approaches for 5G and beyond, covering key performance parameters like ECC, TARC, MEG, and isolation. Discusses wideband, ultra-wideband, multiband, and circular polarized MIMO antennas, along with techniques for improving port isolation.	Provides in-depth insights into the evolution and design considerations of MIMO an- tennas for 5G, emphasiz- ing the superiority of MIMO over traditional SISO sys- tems.	The review may not adequately address the integration challenges of these advanced antenna designs into diverse and existing network infrastructures, particularly in terms of spatial constraints and the electromagnetic compatibility with other system components.

TABLE 4.	Overview of	research on	massive M	IIMO and	related	technologies	in 5G i	networks	(part 3)).
----------	-------------	-------------	-----------	----------	---------	--------------	---------	----------	----------	----

Citation	Method	Explanation	Advantages	Limitations
[53]	MIMO Antenna Design and Applications	Covers the fundamentals, performance parameters, design approaches, and methodologies for MIMO antennas, focusing on UWB, dual-band, and circularly polarized antennas. Discusses isolation techniques, diversity parameters, and the impact of MIMO antennas in indoor environments and future 6G technology.	Provides a detailed review of MIMO antenna designs, of- fering insights into achiev- ing high data rates, reliabil- ity, and strong transmission quality for modern commu- nication systems. Highlights the importance of MIMO an- tennas in enhancing chan- nel capacity and supporting a wide range of applications.	Limited Scope on Emerging MIMO Technologies. The paper not extensively cover newer MIMO technologies and configurations, limiting its applicability to future wireless communication advancements.
[85]	Hybrid Beamforming in Massive MIMO	Evaluates the performance of analog, digital, and hybrid beamforming techniques in massive MIMO systems, highlighting the hybrid approach which combines elements of both analog and digital beamforming to reduce complexity and cost.	Offers a balance between the flexibility of digital beam- forming and the energy ef- ficiency of analog beam- forming. Suitable for mas- sive MIMO configurations, enhancing data rates and net- work efficiency with fewer RF chains.	The complexity of hybrid beamforming algorithms can still be significant, especially in systems with a very large number of antennas. The per- formance might not fully match that of pure digital beamforming in certain sce- narios.
[86]	Deep Learning-Based Beam- forming	This paper proposes a novel deep learning-based beamforming algorithm for optimizing beamformers in massive MIMO networks, particularly focusing on the weighted sum rate maximization problem in heterogeneous environments.	Addresses the computational complexity associated with traditional optimization algorithms. The method is designed to generalize well across heterogeneous network scenarios, including variations in the number of antennas and base stations.Demonstrates high weighted sum rates with significantly reduced runtime.	The method's performance and generalization capabili- ties might be sensitive to the training data quality and di- versity, potentially limiting its effectiveness in scenarios vastly different from those in the training set.
[87]	Deep Learning-Based Coor- dinated Beamforming	Utilizes a deep learning model to predict effective beamforming vectors for mmWave systems, addressing challenges such as narrow beam usage, signal blockage, and frequent hand-offs in highly mobile environments.	Enhances coverage and reliability in highly mobile mmWave networks. Reduces training overhead for beamforming vector determination. Demonstrates superior attainable rates compared to traditional beamforming methods.	May require extensive data for training the deep learn- ing model, potentially limit- ing rapid deployment in dy- namically changing environ- ments.
[88]	Hybrid Beamforming in MU-mMIMO Systems	Designs a downlink MU- mMIMO communication system at mmWave frequencies with multiple data streams per user and accurate CSI, focusing on hybrid precoding at the transmitter and combining at the receiver.	Increases spectral efficiency by leveraging large antenna arrays and coherent processing. Addresses mmWave high path-loss through higher antenna gains. Demonstrates the feasibility and benefits of mmWave and mMIMO in 5G NR as per 3GPP. Explores the tradeoff between multiple data streams per user and the number of BS antennas, advocating for more parallel data streams to boost throughput.	Complexity of hybrid beam- forming design and imple- mentation. Requirement for accurate CSI, which can be challenging in dynamic envi- ronments. Tradeoff analysis may oversimplify real-world deployment challenges.

traffic generated by these advanced antenna systems without becoming a bottleneck is a significant challenge.

15. Deployment in Varied Environments: Designing scalable modular antenna arrays that can be efficiently deployed in diverse environments, from densely populated urban areas to rural regions, presents unique challenges in terms of coverage, performance, and installation logistics.

16. Synchronization and Timing Accuracy: In Massive MIMO systems, precise synchronization and timing accuracy are essential for coherent signal transmission and reception across multiple antennas. Achieving this level of synchronization in a scalable system is technically challenging.

17. Adaptability to Standards and Regulations: Ensuring that antenna systems are adaptable to evolving industry standards and comply with regulatory requirements, especially in terms of electromagnetic emissions and spectrum usage, is a technical and operational challenge.

18. Hardware Impairments and Non-Reciprocal Transceivers: Addressing hardware impairments and the challenges posed by non-reciprocal transceivers is essential. These factors can contribute to pilot contamination and affect the accuracy of channel state information (CSI), crucial for network performance.

19. Spectral and Energy Efficiency: Balancing the spectral and energy efficiency of antenna systems, especially in the context of environmental sustainability and operational costs, remains a critical challenge. Developing energy-efficient antenna designs without compromising on performance is key.

20. Implementation and Maintenance: The scalability of modular antenna arrays implies ease of implementation and maintenance. Ensuring that these systems are not only easy to deploy but also maintain and upgrade over time, without causing significant network disruptions, is essential for long-term viability.

21. User Demand and Network Traffic Management: As user demand and network traffic continue to grow, managing this traffic efficiently while maintaining high quality of service is a significant challenge. Scalable modular antenna arrays must be capable of handling increased loads without degradation in performance.

The challenges in scalable modular antenna arrays for 5G Massive MIMO networks, while significant, provide clear pathways for future research and development.

1. Advanced Signal Processing Algorithms: Future research should focus on developing more sophisticated signal processing algorithms that are capable of handling the increased complexity of Massive MIMO systems efficiently. These algorithms should aim to optimize system performance while minimizing computational overhead.

2. Dynamic Beamforming Techniques: There is a need for advanced dynamic beamforming techniques that can adapt in real-time to changing environmental conditions and user mobility. Research should aim to develop algorithms that can quickly adjust beam patterns for optimal network performance. 3. Enhanced CSI Acquisition Methods: Developing more robust methods for CSI acquisition in dynamic environments is crucial. Future work could explore machine learning and AI-based approaches for predictive and adaptive CSI acquisition.

4. Scalable System Design: Research should continue to focus on designing antenna systems that can balance scalability with performance. This includes exploring modular designs that allow for easy expansion and upgrading of network capabilities.

5. Interference Management Solutions: Developing new techniques for managing inter-cell interference in dense network environments is essential. This includes research into spatial filtering and network coordination methods.

6. Innovative Hardware Design: Future directions should include innovative hardware designs that address limitations such as size, weight, and thermal management, and improve calibration processes for large-scale antenna arrays.

7. Spectrum Efficiency Optimization: Exploring ways to optimize spectrum utilization while maintaining high energy efficiency will be a key area of future research. This includes developing new antenna technologies and signal processing techniques.

8. High-Capacity Backhaul Solutions: As data rates continue to increase, developing high-capacity backhaul solutions that can support the massive data traffic generated by 5G networks will be crucial.

9. Versatile Deployment Strategies: Research into versatile deployment strategies for varied environments, from urban to rural settings, is needed. This includes developing antenna systems that are adaptable to different installation scenarios.

10. Precise Synchronization Techniques: Future work should focus on enhancing synchronization and timing accuracy in Massive MIMO systems, especially as the scale of these systems increases.

11. Compliance with Evolving Standards: Ongoing research is needed to ensure that antenna systems remain adaptable to evolving industry standards and comply with regulatory requirements.

12. Leveraging AI and Machine Learning: Utilizing AI and machine learning for predictive network management, automated optimization, and real-time decision-making could be a significant area of future development.

13. Cross-Disciplinary Collaboration: Collaborations between academia, industry, and regulatory bodies will be vital in addressing the multifaceted challenges of 5G Massive MIMO networks and steering the direction of future research.

VIII. CONCLUSION

The exploration of scalable modular antenna arrays in 5G Massive MIMO networks has highlighted a landscape rich with both challenges and opportunities. As we stand at the cusp of a new era in wireless communication, it is evident that the successful deployment and optimization of 5G networks hinge on overcoming a series of technical, operational, and commercial challenges. From the integration

complexities of advanced antenna systems into existing network infrastructures to the intricacies of signal processing, beamforming, and channel state information (CSI) acquisition, each challenge presents a unique obstacle. However, these obstacles also open doors to innovative solutions. The development of more efficient signal processing algorithms, dynamic beamforming techniques, and robust methods for CSI acquisition is crucial for enhancing network performance and capacity.

The commercial aspect, including the rapid adoption and adaptation of MIMO technology by leading companies, underscores the significant commercial potential of 5G technology. The evolution from the first commercial MIMO systems to the current state-of-the-art designs by companies like Huawei, Ericsson, and Qualcomm highlights a trajectory of continuous innovation and adaptation. Furthermore, the need for scalable system designs that balance performance with complexity, the management of inter-cell interference, and the quest for spectrum efficiency optimization continue to drive research and development in this field. The challenges of hardware limitations, thermal management, and backhaul capacity underscore the need for a multifaceted approach that encompasses advanced technological development, strategic planning, and cross-disciplinary collaboration.

Looking to the future, leveraging AI and machine learning for network management, focusing on precise synchronization techniques, and ensuring compliance with evolving standards and regulations are areas ripe for exploration. The direction of future research and development is clear – it must be innovative, interdisciplinary, and inclusive of evolving commercial and technical realities. To conclude, the journey towards realizing the full potential of 5G Massive MIMO networks is complex and multifaceted. It requires a concerted effort from academia, industry, and regulatory bodies to address the challenges and harness the opportunities presented by this revolutionary technology. The path forward is paved with challenges, but it is these challenges that will drive the innovations necessary to usher in the next wave of advancements in wireless communication.

ACKNOWLEDGMENT

The authors would like to express their sincere thanks to the Kulliyyah of Engineering at the International Islamic University Malaysia for awarding the IIUM Engineering Merit Scholarship to the Arselan Ashraf and for providing a supportive and conducive environment for their research activities.

REFERENCES

- R. Chataut and R. Akl, "Massive MIMO systems for 5G and beyond networks—Overview, recent trends, challenges, and future research direction," *Sensors*, vol. 20, no. 10, p. 2753, May 2020, doi: 10.3390/s20102753.
- [2] M. Giordani, M. Polese, M. Mezzavilla, S. Rangan, and M. Zorzi, "Toward 6G networks: Use cases and technologies," *IEEE Commun. Mag.*, vol. 58, no. 3, pp. 55–61, Mar. 2020, doi: 10.1109/MCOM.001.1900411.
- [3] X. Chen, D. W. K. Ng, W. Yu, E. G. Larsson, N. Al-Dhahir, and R. Schober, "Massive access for 5G and beyond," *IEEE J. Sel. Areas Commun.*, vol. 39, no. 3, pp. 615–637, Mar. 2021, doi: 10.1109/JSAC.2020.3019724.

- [4] N. H. M. Adnan, I. Md. Rafiqul, and A. H. M. Z. Alam, "Massive MIMO for fifth generation (5G): Opportunities and challenges," in *Proc. Int. Conf. Comput. Commun. Eng. (ICCCE)*, Jul. 2016, pp. 47–52, doi: 10.1109/ICCCE.2016.23.
- [5] P. De and S. Singh, "Journey of mobile generation and cognitive radio technology in 5G," *Int. J. Mobile Netw. Commun. Telematics*, vol. 6, nos. 4–6, pp. 1–19, Dec. 2016, doi: 10.5121/ijmnct.2016.6601.
- [6] M. S. Sharawi, "Advancements in MIMO antenna systems," in *Developments in Antenna Analysis and Design*, vol. 1, R. Mittra, Ed. Montreal, QC Canada: Institution of Engineering and Technology, 2018, pp. 109–127.
- [7] A. Gohil, H. Modi, and S. K. Patel, "5G technology of mobile communication: A survey," in *Proc. Int. Conf. Intell. Syst. Signal Process.* (*ISSP*), Mar. 2013, pp. 288–292, doi: 10.1109/ISSP.2013.6526920.
- [8] K. Muppavaram, S. Govathoti, D. Kamidi, and T. Bhaskar, "Exploring the generations: A comparative study of mobile technology from 1G to 5G," *Int. J. Electron. Commun. Eng.*, vol. 10, no. 7, pp. 54–62, Jul. 2023, doi: 10.14445/23488549/ijece-v10i7p106.
- [9] L. J. Vora, "Evolution of mobile generation technology: 1G to 5G and review of upcoming wireless technology 5G," *Int. J. Mod. Trends Eng. Res.*, vol. 2, no. 10, pp. 281–290, 2015. [Online]. Available: https://api.semanticscholar.org/CorpusID:31229954
- [10] A. Biswas and V. R. Gupta, "Multiband antenna design for smartphone covering 2G, 3G, 4G and 5G NR frequencies," in *Proc. 3rd Int. Conf. Trends Electron. Informat. (ICOEI)*, Apr. 2019, pp. 84–87, doi: 10.1109/ICOEI.2019.8862713.
- [11] E. Ezhilarasan and M. Dinakaran, "A review on mobile technologies: 3G, 4G and 5G," in *Proc. 2nd Int. Conf. Recent Trends Challenges Comput. Models (ICRTCCM)*, Feb. 2017, pp. 369–373, doi: 10.1109/ICRTCCM.2017.90.
- [12] V. R. Rentapalli and Z. J. Khan, "MIMO and smart antenna technologies for 3G and 4G," in *Information Technology and Mobile Communication*. Berlin, Germany: Springer-Verlag, 2011, pp. 493–498, doi: 10.1007/978-3-642-20573-6-90.
- [13] U. Varshney, "4G wireless networks," *IT Prof.*, vol. 14, no. 5, pp. 34–39, Sep. 2012, doi: 10.1109/MITP.2012.71.
- [14] J. Zhang, Z. Wu, W. Xie, and F. Yang, "MEC architectures in 4G and 5G mobile networks," in *Proc. 10th Int. Conf. Wireless Commun. Signal Process. (WCSP)*, Oct. 2018, pp. 1–5, doi: 10.1109/WCSP.2018.8555652.
- [15] T. R. Kumar and M. Moorthi, "Review on 4G antenna design for LTE application," in *Proc. 3rd Int. Conf. Adv. Electr., Electron, Inf., Commun. Bio-Informatics (AEEICB)*, Feb. 2017, pp. 476–478, doi: 10.1109/AEEICB.2017.7972357.
- [16] M. Rumney, P. Cain, T. Barratt, A. L. Freire, W. Yuan, E. Mellios, and M. Beach, "Testing 5G: Evolution or revolution?" in *Proc. Radio Propag. Technol. 5G*, Oct. 2016, pp. 1–9, doi: 10.1049/ic.2016.0067.
- [17] D. Turkalj, "Effect of 5G network on development of digitally dependent industries," in *Proc. 44th Int. Conv. Inf., Commun. Electron. Technol. (MIPRO)*, Sep. 2021, pp. 1353–1357, doi: 10.23919/MIPRO52101.2021.9597107.
- [18] P. Tiwari, V. Gahlaut, M. Kaushik, P. Rani, A. Shastri, and B. Singh, "Advancing 5G connectivity: A comprehensive review of MIMO antennas for 5G applications," *Int. J. Antennas Propag.*, vol. 2023, pp. 1–19, Aug. 2023, doi: 10.1155/2023/5906721.
- [19] E. Björnson, J. Hoydis, and L. Sanguinetti, "Massive MIMO networks: Spectral, energy, and hardware efficiency," *Found. Trends Signal Process.*, vol. 11, nos. 3–4, pp. 154–655, 2017, doi: 10.1561/200000093.
- [20] N. Hassan and X. Fernando, "Massive MIMO wireless networks: An overview," *Electronics*, vol. 6, no. 3, p. 63, Sep. 2017, doi: 10.3390/electronics6030063.
- [21] K. N. R. S. V. Prasad, E. Hossain, and V. K. Bhargava, "Energy efficiency in massive MIMO-based 5G networks: Opportunities and challenges," *IEEE Wireless Commun.*, vol. 24, no. 3, pp. 86–94, Jun. 2017, doi: 10.1109/MWC.2016.1500374WC.
- [22] T. E. Bogale and L. B. Le, "Massive MIMO and mmWave for 5G wireless HetNet: Potential benefits and challenges," *IEEE Veh. Technol. Mag.*, vol. 11, no. 1, pp. 64–75, Mar. 2016, doi: 10.1109/MVT.2015.2496240.
- [23] M. A. Albreem, M. Juntti, and S. Shahabuddin, "Massive MIMO detection techniques: A survey," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 4, pp. 3109–3132, 4th Quart., 2019, doi: 10.1109/COMST.2019.2935810.
- [24] H. Chen and R. Gantile, "Spatial multiplexing for 5G wireless communications," *Microw. J. Int. Ed.*, vol. 61, pp. 114–126, Feb. 2018.
- [25] M. Benzaghta and K. M. Rabie, "Massive MIMO systems for 5G: A systematic mapping study on antenna design challenges and channel estimation open issues," *IET Commun.*, vol. 15, no. 13, pp. 1677–1690, Aug. 2021, doi: 10.1049/cmu2.12180.

- [26] E. Ali, M. Ismail, R. Nordin, and N. F. Abdulah, "Beamforming techniques for massive MIMO systems in 5G: Overview, classification, and trends for future research," *Frontiers Inf. Technol. Electron. Eng.*, vol. 18, no. 6, pp. 753–772, Jun. 2017, doi: 10.1631/fitee.1601817.
- [27] Z. Zhang, Z. Jiang, B. Yang, and X. She, "A beamforming-based enhanced handover scheme with adaptive threshold for 5G heterogeneous networks," *Electronics*, vol. 12, no. 19, p. 4131, Oct. 2023, doi: 10.3390/electronics12194131.
- [28] I. Kumar, V. Sachan, R. Shankar, and R. K. Mishra, "Performance analysis of multi-user massive MIMO systems with perfect and imperfect CSI," *Proc. Comput. Sci.*, vol. 167, pp. 1452–1461, Jan. 2020, doi: 10.1016/j.procs.2020.03.356.
- [29] T. R. Delson and I. Jose, "Study on 5G massive MIMO technology key parameters for spectral efficiency improvement including SINR mapping on rural area test case," in *Proc. IEEE 3rd Global Conf. Advancement Technol. (GCAT)*, Oct. 2022, pp. 1–6, doi: 10.1109/GCAT55367.2022.9972156.
- [30] T. Kim and S. Park, "Statistical beamforming for massive MIMO systems with distinct spatial correlations," *Sensors*, vol. 20, no. 21, p. 6255, Nov. 2020, doi: 10.3390/s20216255.
- [31] M. Nguyen, "Massive MIMO: A survey of benefits and challenges," *ICSES Trans. Comput. Hardw. Electr. Eng.*, vol. 4, no. 4, pp. 1–4, 2018.
- [32] E. Björnson, J. Hoydis, and L. Sanguinetti, "Massive MIMO has unlimited capacity," *IEEE Trans. Wireless Commun.*, vol. 17, no. 1, pp. 574–590, Jan. 2018, doi: 10.1109/TWC.2017.2768423.
- [33] Y. Liu, B. Ai, and J. Zhang, "Downlink spectral efficiency of massive MIMO systems with mutual coupling," *Electronics*, vol. 12, no. 6, p. 1364, Mar. 2023, doi: 10.3390/electronics12061364.
- [34] H. Halbauer, A. Weber, D. Wiegner, and T. Wild, "Energy efficient massive MIMO array configurations," in *Proc. IEEE GLOBECOM Workshops (GC Wkshps)*, Dec. 2018, pp. 1–6, doi: 10.1109/GLOCOMW.2018.8644331.
- [35] L. Lu, G. Y. Li, A. L. Swindlehurst, A. Ashikhmin, and R. Zhang, "An overview of massive MIMO: Benefits and challenges," *IEEE J. Sel. Topics Signal Process.*, vol. 8, no. 5, pp. 742–758, Oct. 2014, doi: 10.1109/JSTSP.2014.2317671.
- [36] A. Ashikhmin, L. Li, and T. L. Marzetta, "Interference reduction in multi-cell massive MIMO systems with large-scale fading precoding," *IEEE Trans. Inf. Theory*, vol. 64, no. 9, pp. 6340–6361, Sep. 2018, doi: 10.1109/TIT.2018.2853733.
- [37] T. Zemen, D. Löschenbrand, M. Hofer, C. Pacher, and B. Rainer, "Orthogonally precoded massive MIMO for high mobility scenarios," *IEEE Access*, vol. 7, pp. 132979–132990, 2019, doi: 10.1109/ACCESS.2019.2941316.
- [38] Y. Huo, X. Lin, B. Di, H. Zhang, F. J. L. Hernando, A. S. Tan, S. Mumtaz, Ö. T. Demir, and K. Chen-Hu, "Technology trends for massive MIMO towards 6G," *Sensors*, vol. 23, no. 13, p. 6062, Jun. 2023, doi: 10.3390/s23136062.
- [39] S. Willhammar, J. Flordelis, L. Van der Perre, and F. Tufvesson, "Channel hardening in massive MIMO—A measurement based analysis," in *Proc. IEEE 19th Int. Workshop Signal Process. Adv. Wireless Commun.* (SPAWC), Jun. 2018, pp. 1–5, doi: 10.1109/SPAWC.2018.8445925.
- [40] M. Pons, E. Valenzuela, B. Rodríguez, J. A. Nolazco-Flores, and C. Del-Valle-Soto, "Utilization of 5G technologies in IoT applications: Current limitations by interference and network optimization difficulties—A review," *Sensors*, vol. 23, no. 8, p. 3876, Apr. 2023, doi: 10.3390/s23083876.
- [41] S. Buzzi, I. Chih-Lin, T. E. Klein, H. V. Poor, C. Yang, and A. Zappone, "A survey of energy-efficient techniques for 5G networks and challenges ahead," *IEEE J. Sel. Areas Commun.*, vol. 34, no. 4, pp. 697–709, Apr. 2016, doi: 10.1109/JSAC.2016.2550338.
- [42] X. Gu, D. Liu, Y. Hasegawa, K. Masuko, C. Baks, Y. Suto, Y. Fujisaku, B. Sadhu, A. Paidimarri, N. Guan, and A. Valdes-Garcia, "Antennain-package integration for a wideband scalable 5G millimeter-wave phased-array module," *IEEE Microw. Wireless Compon. Lett.*, vol. 31, no. 6, pp. 682–684, Jun. 2021, doi: 10.1109/LMWC.2021. 3071917.
- [43] H. I. Obakhena, A. L. Imoize, F. I. Anyasi, and K. V. N. Kavitha, "Application of cell-free massive MIMO in 5G and beyond 5G wireless networks: A survey," *J. Eng. Appl. Sci.*, vol. 68, no. 1, p. 13, Dec. 2021, doi: 10.1186/s44147-021-00014-y.
- [44] G. Interdonato, Cell-Free Massive MIMO: Scalability, Signal Processing and Power Control, vol. 2090. Linköping, Sweden: Linköping Univ. Electronic Press, 2020, doi: 10.3384/diss.diva-167218.

- [45] S. Sachan, R. Sharma, and A. Sehgal, "Energy efficiency and scalability of 5G networks for IoT in mobile wireless sensor networks," in 5G and Beyond (Springer Tracts in Electrical and Electronics Engineering). Berlin, Germany: Springer, 2023, pp. 151–168, doi: 10.1007/978-981-99-3668-7-8.
- [46] A. Puglielli, N. Narevsky, P. Lu, T. Courtade, G. Wright, B. Nikolic, and E. Alon, "A scalable massive MIMO array architecture based on common modules," in *Proc. IEEE Int. Conf. Commun. Workshop (ICCW)*, Jun. 2015, pp. 1310–1315, doi: 10.1109/ICCW.2015.7247359.
- [47] X. He, Y. Cui, and M. M. Tentzeris, "Tile-based massively scalable MIMO and phased arrays for 5G/B5G-enabled smart skins and reconfigurable intelligent surfaces," *Sci. Rep.*, vol. 12, no. 1, p. 2741, Feb. 2022, doi: 10.1038/s41598-022-06096-9.
- [48] S. Shahramian, M. J. Holyoak, A. Singh, and Y. Baeyens, "A fully integrated 384-element, 16-tile, W-band phased array with self-alignment and self-test," *IEEE J. Solid-State Circuits*, vol. 54, no. 9, pp. 2419–2434, Sep. 2019, doi: 10.1109/JSSC.2019.2928694.
- [49] A. Bhatta, J. Park, D. Baek, and J.-G. Kim, "A multimode 28 GHz CMOS fully differential beamforming IC for phased array transceivers," *Sensors*, vol. 23, no. 13, p. 6124, Jul. 2023, doi: 10.3390/s23136124.
- [50] B. Santhikiran and T. Kavitha, "UWB microstrip fed 4-element MIMO antenna for 5G applications," *Int. J. Intell. Syst. Appl. Eng.*, vol. 11, no. 9s, pp. 342–350, 2023.
- [51] W. Chen, M. M. Tentzeris, Y. Yao, Y. Zhang, and L. Yang, "MIMO antenna design and channel modeling," *Int. J. Antennas Propag.*, vol. 2012, pp. 1–2, Nov. 2012, doi: 10.1155/2012/570718.
- [52] P. Sharma, R. N. Tiwari, P. Singh, P. Kumar, and B. K. Kanaujia, "MIMO antennas: Design approaches, techniques and applications," *Sensors*, vol. 22, no. 20, p. 7813, Oct. 2022, doi: 10.3390/s22207813.
- [53] B. Panzner, W. Zirwas, S. Dierks, M. Lauridsen, P. Mogensen, K. Pajukoski, and D. Miao, "Deployment and implementation strategies for massive MIMO in 5G," in *Proc. IEEE GLOBECOM Workshops* (*GC Wkshps*), Dec. 2014, pp. 346–351, doi: 10.1109/GLOCOMW. 2014.7063455.
- [54] W. M. Abdulkawi, M. A. Alqaisei, A.-F.-A. Sheta, and I. Elshafiey, "New compact antenna array for MIMO Internet of Things applications," *Micromachines*, vol. 13, no. 9, p. 1481, Sep. 2022, doi: 10.3390/mi13091481.
- [55] R. Mayo and S. Harmer, "A cost-effective modular phased array," in *Proc. IEEE Int. Symp. Phased Array Syst. Technol.*, Oct. 2013, pp. 93–96, doi: 10.1109/ARRAY.2013.6731807.
- [56] S. T. Serkan and C. Baktir, "Reliability modeling & analysis for active phased array antenna design," in *Proc. Annu. Rel. Maintainability Symp.* (*RAMS*), 2017, pp. 1–5, doi: 10.1109/RAM.2017.7889764.
- [57] S. Ghosh and D. Sen, "An inclusive survey on array antenna design for millimeter-wave communications," *IEEE Access*, vol. 7, pp. 83137–83161, 2019, doi: 10.1109/ACCESS.2019.2924805.
- [58] K. M. Younus, A. A. Jasim, and R. W. Clarke, "A beam steering system design based on phased array antennas," *IOP Conf. Ser., Mater. Sci. Eng.*, vol. 1152, no. 1, May 2021, Art. no. 012002, doi: 10.1088/1757-899x/1152/1/012002.
- [59] S. Buzzi, C. D'Andrea, and M. Lops, "Using massive MIMO arrays for joint communication and sensing," in *Proc. 53rd Asilomar Conf. Signals, Syst., Comput.*, Nov. 2019, pp. 5–9, doi: 10.1109/IEEECONF44664.2019.9048857.
- [60] Y. Liu, X. Liu, X. Mu, T. Hou, J. Xu, M. Di Renzo, and N. Al-Dhahir, "Reconfigurable intelligent surfaces: Principles and opportunities," *IEEE Commun. Surveys Tuts.*, vol. 23, no. 3, pp. 1546–1577, 3rd Quart., 2021, doi: 10.1109/COMST.2021.3077737.
- [61] M. Lashab, M. Belattar, S. Hocine, and H. Ahmed, "Metamaterial applications in modern antennas," in *Metamaterials—History, Current State, Applications, and Perspectives*. London, U.K.: IntechOpen, 2023, doi: 10.5772/intechopen.108267.
- [62] M. Heikkilä, T. Kippola, P. Kärsämä, A. Nykänen, P. Tuuttila, and M. Matinmikko, "Active antenna system (AAS) capabilities for 5G systems: A field study of performance," in *Proc. 1st Int. Conf. 5G Ubiquitous Connectivity*, Nov. 2014, pp. 181–186, doi: 10.4108/icst.5gu.2014.258113.
- [63] J. Anguera, A. Andújar, J. Jayasinghe, V. V. S. S. S. Chakravarthy, P. S. R. Chowdary, J. L. Pijoan, T. Ali, and C. Cattani, "Fractal antennas: An historical perspective," *Fractal Fractional*, vol. 4, no. 1, p. 3, Jan. 2020, doi: 10.3390/fractalfract4010003.
- [64] L. Wang, J. Yu, T. Xie, and K. Bi, "A novel multiband fractal antenna for wireless application," *Int. J. Antennas Propag.*, vol. 2021, pp. 1–9, Jun. 2021, doi: 10.1155/2021/9926753.
- [65] Lens.Org. Accessed: Dec. 23, 2023. [Online]. Available: https://www.lens. org/lens/search/scholar/listq=massivemimoin5g

- [66] G. Xu, T. Xiao, T. Tao, D. Zhang, W. Li, and H. Ma, "Research on intelligent optimization of massive MIMO coverage based on 5G MR," in Proc. IEEE Int. Conf. Parallel Distrib. Process. Appl., Big Data Cloud Comput., Sustain. Comput. Commun., Social Comput. Netw. (ISPA/BDCloud/SocialCom/SustainCom), Dec. 2020, pp. 1455–1459, doi: 10.1109/ISPA-BDCloud-SocialCom-SustainCom51426.2020.00218.
- [67] R. Polus, A. H. A. El-Malek, and M. Elsabrouty, "User association in limited quota het-nets aided massive MIMO networks," *IEEE Wireless Commun. Lett.*, vol. 10, no. 7, pp. 1366–1369, Jul. 2021, doi: 10.1109/LWC.2020.3039422.
- [68] F. Kasem, K. Rambabu, A. K. Iyer, and W. A. Krzymien, "A reducedcomplexity load-modulated MIMO transmitter readily scalable in 5G massive MIMO transmitters," *IEEE Trans. Circuits Syst. II, Exp. Briefs*, vol. 70, no. 8, pp. 2849–2853, Aug. 2023, doi: 10.1109/TCSII.2023.3250665.
- [69] M. Sakai, K. Kamohara, H. Iura, H. Nishimoto, K. Ishioka, Y. Murata, M. Yamamoto, A. Okazaki, N. Nonaka, S. Suyama, J. Mashino, A. Okamura, and Y. Okumura, "Experimental field trials on MU-MIMO transmissions for high SHF wide-band massive MIMO in 5G," *IEEE Trans. Wireless Commun.*, vol. 19, no. 4, pp. 2196–2207, Apr. 2020, doi: 10.1109/TWC.2019.2962766.
- [70] S. Sun, B. Rong, R. Q. Hu, and Y. Qian, "Spatial domain management and massive MIMO coordination in 5G SDN," *IEEE Access*, vol. 3, pp. 2238–2251, 2015, doi: 10.1109/ACCESS.2015.2498609.
- [71] S. K. Goudos, P. D. Diamantoulakis, and G. K. Karagiannidis, "Multiobjective optimization in 5G wireless networks with massive MIMO," *IEEE Commun. Lett.*, vol. 22, no. 11, pp. 2346–2349, Nov. 2018, doi: 10.1109/LCOMM.2018.2868663.
- [72] E. L. Bengtsson, F. Rusek, S. Malkowsky, F. Tufvesson, P. C. Karlsson, and O. Edfors, "A simulation framework for multiple-antenna terminals in 5G massive MIMO systems," *IEEE Access*, vol. 5, pp. 26819–26831, 2017, doi: 10.1109/ACCESS.2017.2775210.
- [73] S. Dahiya and R. Pal, "L-shape array based technique to reduce cross user correlation in massive MIMO systems," *IEEE Wireless Commun. Lett.*, vol. 12, no. 9, pp. 1628–1631, Sep. 2023, doi: 10.1109/LWC.2023.3285274.
- [74] H. Huang, "A decoupling method for antennas with different frequencies in 5G massive MIMO application," *IEEE Access*, vol. 8, pp. 140273–140278, 2020, doi: 10.1109/ACCESS.2020.3012665.
- [75] O. Elijah, C. Y. Leow, T. A. Rahman, S. Nunoo, and S. Z. Iliya, "A comprehensive survey of pilot contamination in massive MIMO— 5G system," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 2, pp. 905–923, 2nd Quart., 2016, doi: 10.1109/COMST.2015.2504379.
- [76] M. H. Siddiqui, K. Khurshid, I. Rashid, A. A. Khan, and K. Ahmed, "Optimal massive MIMO detection for 5G communication systems via hybrid n-bit heuristic assisted-VBLAST," *IEEE Access*, vol. 7, pp. 173646–173656, 2019, doi: 10.1109/ACCESS.2019.2949247.
- [77] Y. Zhang, X. Li, L. Pang, Y. He, G. Ren, and J. Li, "A 2-D geometrybased stochastic channel model for 5G massive MIMO communications in real propagation environments," *IEEE Syst. J.*, vol. 15, no. 1, pp. 307–318, Mar. 2021, doi: 10.1109/JSYST.2020.2971062.
- [78] Y. Li, C.-Y.-D. Sim, Y. Luo, and G. Yang, "12-port 5G massive MIMO antenna array in sub-6 GHz mobile handset for LTE bands 42/43/46 applications," *IEEE Access*, vol. 6, pp. 344–354, 2018, doi: 10.1109/ACCESS.2017.2763161.
- [79] X. Zeng, F. Zhang, B. Wang, and K. J. R. Liu, "Radio frequency based direction sensing using massive MIMO," *IEEE Access*, vol. 8, pp. 26827–26838, 2020, doi: 10.1109/ACCESS.2020.2964525.
- [80] S. Chen, J. Zhang, E. Björnson, J. Zhang, and B. Ai, "Structured massive access for scalable cell-free massive MIMO systems," *IEEE J. Sel. Areas Commun.*, vol. 39, no. 4, pp. 1086–1100, Apr. 2021, doi: 10.1109/JSAC.2020.3018836.
- [81] H. Jin, K. Liu, M. Zhang, L. Zhang, G. Lee, E. N. Farag, D. Zhu, E. Onggosanusi, M. Shafi, and H. Tataria, "Massive MIMO evolution toward 3GPP release 18," *IEEE J. Sel. Areas Commun.*, vol. 41, no. 6, pp. 1635–1654, Jun. 2023, doi: 10.1109/JSAC.2023.3273768.
- [82] S. Senger and P. K. Malik, "A comprehensive survey of massive-MIMO based on 5G antennas," *Int. J. RF Microw. Comput.-Aided Eng.*, vol. 32, no. 12, 2022, Art. no. e23496, doi: 10.1002/mmce.23496.
- [83] S. S. Al-Bawri, M. T. Islam, M. S. Islam, M. J. Singh, and H. Alsaif, "Massive metamaterial system-loaded MIMO antenna array for 5G base stations," *Sci. Rep.*, vol. 12, no. 1, p. 14311, Aug. 2022, doi: 10.1038/s41598-022-18329-y.
- [84] T. Raj, R. Mishra, P. Kumar, and A. Kapoor, "Advances in MIMO antenna design for 5G: A comprehensive review," *Sensors*, vol. 23, no. 14, p. 6329, Jul. 2023, doi: 10.3390/s23146329.

- [85] S. Hamid, S. R. Chopra, A. Gupta, S. Tanwar, B. C. Florea, D. D. Taralunga, O. Alfarraj, and A. M. Shehata, "Hybrid beamforming in massive MIMO for next-generation communication technology," *Sensors*, vol. 23, no. 16, p. 7294, Aug. 2023, doi: 10.3390/s23167294.
- [86] M. Zhu, T.-H. Chang, and M. Hong, "Learning to beamform in heterogeneous massive MIMO networks," *IEEE Trans. Wireless Commun.*, vol. 22, no. 7, pp. 4901–4915, Jul. 2023, doi: 10.1109/TWC.2022.3230662.
- [87] R. R. Kumar, "Beamforming in massive MIMO antenna arrays for 5G," SSRN J., pp. 1–12, May 2023, doi: 10.2139/ssrn.4439290.
- [88] R. Dilli, "Performance analysis of multi user massive MIMO hybrid beamforming systems at millimeter wave frequency bands," *Wireless Netw.*, vol. 27, no. 3, pp. 1925–1939, Apr. 2021, doi: 10.1007/s11276-021-02546-w.
- [89] Wireless Technology & Innovation Mobile Technology, Qualcomm, Chennai, India, 2022.
- [90] NIHF. Inductee Arogyaswami Paulraj Invented MIMO Technology. Accessed: Jan. 3, 2024. [Online]. Available: https://www.invent.org/ blog/inventors/Arogyaswami-Paulraj
- [91] Shenzhen VLG Wireless Technology Co Ltd. Accessed: Jan. 3, 2024.[Online]. Available: https://www.vlgcom.com/
- [92] Chinmore Industry Co Ltd. Accessed: Jan. 3, 2024. [Online]. Available: http://www.chinmore.com.tw/
- [93] *RÉMO Electronics*. Accessed: Jan. 3, 2024. [Online]. Available: https://www.remotek.com/
- [94] Airgain Inc. Accessed: Jan. 3, 2024. [Online]. Available: https://www. airgain.com/
- [95] Airspan Networks 5G Solutions. Accessed: Jan. 4, 2024. [Online]. Available: https://www.airspan.com/products/5gsolutions/
- [96] *Huawei 5G Antenna Portfolio*. Accessed: Jan. 4, 2024. [Online]. Available: https://www.huawei.com/en/products/wireless/antennas
- [97] *Taoglas 5G Antennas*. Accessed: Jan. 4, 2024. [Online]. Available: https://www.taoglas.com/products-category/5g-antennas/
- [98] Panorama Antennas 5G Antennas. Accessed: Jan. 4, 2024. [Online]. Available: https://www.panorama-antennas.com/products/5g-antennas
- [99] Ericsson Massive MIMO Antenna. Accessed: Jan. 4, 2024. [Online]. Available: https://www.ericsson.com/en/press-releases/2018/1/ericssonmassive-mimo-now-shipping-to-customers



ARSELAN ASHRAF (Student Member, IEEE) received the Bachelor of Technology (B.Tech.) degree in computer science engineering from Baba Ghulam Shah Badshah University, Rajouri, India, and the Master of Science (M.S.) degree in computer and information engineering from International Islamic University Malaysia, where he is currently pursuing the Ph.D. degree in computer and information engineering. His research interests include machine learning, signal and

image processing, computer vision, and computer network and security.



TEDDY SURYA GUNAWAN (Senior Member, IEEE) received the B.Eng. degree (cum laude) in electrical engineering from Institut Teknologi Bandung (ITB), Indonesia, in 1998, the M.Eng. degree from the School of Computer Engineering, Nanyang Technological University, Singapore, in 2001, and the Ph.D. degree from the School of Electrical Engineering and Telecommunications, University of New South Wales (UNSW), Australia, in 2007. He was a Visiting Research

Fellow with UNSW, from 2010 to 2021, and an Adjunct Professor with Telkom University, from 2022 to 2023. He has been a Professor, since 2019, the Head of the Department of Electrical and Computer Engineering, from 2015 to 2016, and the Head of the Program Accreditation and Quality Assurance, Faculty of Engineering, International Islamic University Malaysia, from 2017 to 2018. His research interests include speech and audio processing, biomedical signal processing and instrumentation, image and video processing, and parallel computing. He was awarded the Best Researcher Award in 2018 from IIUM. He was the Chairperson of the IEEE Instrumentation and Measurement Society–Malaysia Section (2013, 2014, 2020, and 2021). He has been a Chartered Engineer (IET, U.K.), since 2016, and Insinyur Profesional Utama (PII), Indonesia, since 2021, a Registered ASEAN Engineer, since 2018, and an ASEAN Chartered Professional Engineer, since 2020.



MIRA KARTIWI (Member, IEEE) is currently a Professor with the Department of Information Systems, Kulliyyah of Information and Communication Technology, and also the Director of Centre for Professional Development, International Islamic University Malaysia (IIUM). She is an experienced consultant specializing in the health, financial, and manufacturing sectors. Her research interests include health informatics, e-commerce, data mining, information systems

strategy, business process improvement, product development, marketing, delivery strategy, workshop facilitation, training, and communications. She has been appointed as an editorial board member in local and international journals to acknowledge her expertise. She was one of the recipients of Australia Postgraduate Award (APA) in 2004. For her achievement in research, she was awarded the Higher Degree Research Award for Excellence in 2007.



BAMBANG SETIA NUGROHO (Member, IEEE) received the degree in telecommunications from the Department of Electrical Engineering, Institut Teknologi Bandung (ITB), Bandung, Indonesia, in 1999, the master's degree from the Graduate School of Electrical Engineering, ITB, in 2004, and the Ph.D. degree from the Graduate School of Electrical Engineering, Universitas Indonesia, in 2015. Since 1999, he has been a Lecturer and a Researcher with the School of Electrical

Engineering, Telkom University, Indonesia. He has authored more than 25 scientific publications in the telecommunication area. He is a member of the IEEE Communication Society, Antenna and Propagation Society. He has been involved as reviewers for various international conference on telecommunications and actively conducting research in telecommunication and antenna engineering.



RINA PUDJI ASTUTI (Member, IEEE) received the degree in telecommunications from the Department of Electrical Engineering, Institut Teknologi Sepuluh Nopember (ITS), Surabaya, Indonesia, in 1987, and the master's and Ph.D. degrees from the School of Electrical Engineering and Informatics-Telecommunication, Institut Teknologi Bandung (ITB), in 1999 and 2009, respectively. Since 1993, she has been with Telkom University (previously known as Institut

Teknologi Telkom, IT Telkom), Bandung, Indonesia, as a Lecturer. Her research interests include issues related to advanced wireless technology (including massive MIMO, coded OFDM, MC CDMA, and space time code), adaptive transmission scheme, multiuser detection, cooperative and coexistence radio, interference mitigation and bandwidth management, telecommunications engineering and regulation, satellite systems, and radio techniques. She was the Chairperson of IEEE ComSoc Indonesia Chapter from 2016 to 2019.



LEVY OLIVIA NUR (Member, IEEE) received the B.E., M.E., and Ph.D. degrees in electrical engineering, majoring in telecommunication engineering from Institut Teknologi Bandung, Indonesia, in 2001, 2004, and 2014, respectively. From 2002 to 2003, during her study in master's program, she joined Japanese University Studies in Science Technology (JUSST), as a Visiting Student with The University of Electro-Communication (UEC), Tokyo, Japan. She joined

Telkom University (previously known as Institut Teknologi Telkom, IT Telkom), Bandung, Indonesia, as a Lecturer, in 2010. Her research interests include microwave absorbent material, wearable antenna, and its characterization.

. . .